## **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

## **FORM 10-Q**

## QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2015

OR

## □ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission file number: 0-26056

# Image Sensing Systems, Inc. (Exact name of registrant as specified in its charter)

Minnesota

to

(State or other jurisdiction of incorporation or organization)

**500 Spruce Tree Centre 1600 University Avenue West** St. Paul. MN (Address of principal executive offices)

41-1519168 (I.R.S. Employer Identification No.)

(651) 603-7700

(Registrant's telephone number, including area code)

Not Applicable

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ⊠ No 🗆

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes  $\boxtimes$ No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer  $\Box$ 

Non-accelerated filer  $\Box$ (Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 🛛 No 🖂

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Outstanding at August 1, 2015 Class Common Stock, \$0.01 par value per share 5,013,663 shares

Table of Contents

55104

(Zip Code)

Accelerated filer  $\Box$ 

Smaller reporting company  $\boxtimes$ 

## IMAGE SENSING SYSTEMS, INC.

## TABLE OF CONTENTS

PART I. F	INANCIAL INFORMATION	3
<u>Item 1</u> .	Financial Statements (unaudited)	3
Condensed	Consolidated Balance Sheets	3
Condensed	Consolidated Statements of Operations	4
Condensed	Consolidated Statements of Comprehensive Income (Loss)	5
Condensed	Consolidated Statements of Cash Flows	6
Notes to C	ondensed Consolidated Financial Statements	7
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	17
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	25
<u>Item 4.</u>	Controls and Procedures	25
<u>PART II.</u>	OTHER INFORMATION	26
<u>Item 1.</u>	Legal Proceedings	26
Item 1A.	Risk Factors	26
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	26
<u>Item 3.</u>	Defaults Upon Senior Securities	26
<u>Item 4.</u>	Mine Safety Disclosures	26
<u>Item 5.</u>	Other Information	26
<u>Item 6.</u>	Exhibits	26
<u>SIGNATU</u>	JRES	27
EXHIBIT	INDEX	28

## PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

Image Sensing Systems, Inc. Condensed Consolidated Balance Sheets (Unaudited) (in thousands)

		June 30, 2015	De	cember 31, 2014
ASSETS				
Current assets:				
Cash and cash equivalents	\$	575	\$	2,656
Accounts receivable, net of allowance for doubtful accounts of \$106 and \$516, respectively		5,102		4,219
Inventories		2,204		2,234
Prepaid expenses and other current assets		648		871
Total current assets		8,529		9,980
Property and equipment:				
Furniture and fixtures		651		620
Leasehold improvements		575		556
Equipment		3,925		3,964
		5,151		5,140
Accumulated depreciation		4,410		4,279
		741		861
Intangible assets, net		3,228		3,987
Deferred income taxes		61		62
TOTAL ASSETS	\$	12,559	\$	14,890
LIABILITIES AND SHAREHOLDERS' EQUITY				
Current liabilities:				
Accounts payable	\$	3,145	\$	3,315
Warranty and other current liabilities	φ	2,623	φ	2,783
Accrued restructuring		2,023		2,785
Total current liabilities		5,768		6,314
1 otal current haointies		3,708		0,314
Deferred income taxes		166		165
Other long-term liabilities		80		91
Shareholders' equity				
Preferred stock, \$.01 par value; 5,000,000 shares authorized, none issued or outstanding				
Common stock, \$.01 par value; 20,000,000 shares authorized, 5,013,663 and 4,995,963 issued and				
outstanding, respectively		50		49
Additional paid-in capital		23,698		23,547
Accumulated other comprehensive loss		(177)		(158)
Accumulated deficit		(17,026)		(15,118)
Total shareholders' equity		6,545		8,320
TOTAL LIABILITIES AND SHAREHOLDERS' EQUITY	\$	12,559	\$	14,890

See accompanying notes to the condensed consolidated financial statements.

## Image Sensing Systems, Inc. Condensed Consolidated Statements of Operations (Unaudited) (in thousands, except per share data)

	Three-Month Periods Ended June 30,			Six-Month Periods Ended June 30,			
	2015		2014		2015		2014
Revenue:							
Product sales	\$ 2,572	\$	2,651	\$	4,900	\$	4,549
Royalties	 2,609		3,287		4,620		5,711
	5,181		5,938		9,520		10,260
Cost of revenue:							
Product sales	977		1,675		1,996		2,880
	 977		1,675		1,996	-	2,880
Gross profit	 4,204		4,263		7,524		7,380
Operating expenses:							
Selling, marketing and product support	1,322		2,471		2,737		5,191
General and administrative	1,390		1,498		2,762		2,870
Research and development	1,123		1,423		2,200		3,239
Amortization of intangible assets	382		393		763		782
Sale of business unit	751				857		—
Restructuring			—		119		460
Investigation matter	 		36				152
	4,968		5,821		9,438		12,694
Loss from operations	 (764)		(1,558)		(1,914)		(5,314)
Other, net	30		15		29		14
Loss before income taxes	 (734)		(1,543)		(1,885)		(5,300)
Income tax expense (benefit)	8		(10)		24		(10)
Net loss	\$ (742)	\$	(1,533)	\$	(1,909)	\$	(5,290)
Net loss per share:							
Basic	\$ (0.15)	\$	(0.31)	\$	(0.38)	\$	(1.06)
Diluted	\$ (0.15)	\$	(0.31)	\$	(0.38)	\$	(1.06)
Weighted average number of common shares outstanding:	 						
Basic	5 009		4.080		5.002		4.079
	 5,008		4,980		5,003		4,978
Diluted	 5,008	_	4,980		5,003		4,978

See accompanying notes to the condensed consolidated financial statements.

#### Image Sensing Systems, Inc. Condensed Consolidated Statements of Comprehensive Income (Loss) (in thousands)

		Three-Month Periods Ended June 30,				Six-Month Periods Ended June 30,			
	2015		2014		2015		2014		
Loss before income taxes	\$	(742)	\$	(1,533)	\$	(1,909)	\$	(5,290)	
Other comprehensive income (loss):									
Foreign currency translation adjustment		158		184		(19)		215	
Comprehensive loss	\$	(584)	\$	(1,349)	\$	(1,928)	\$	(5,075)	

See accompanying notes to the condensed consolidated financial statements.

## Image Sensing Systems, Inc. Condensed Consolidated Statements of Cash Flows (Unaudited) (in thousands)

		Periods Ended ne 30,	1
	2015	2	2014
Operating activities:			
Net loss	\$ (1,909)	\$	(5,290)
Adjustments to reconcile net loss to net cash used for operating activities:			
Depreciation	173		298
Amortization	763		782
Stock-based compensation	151		186
Loss on disposal of assets	4		
Changes in operating assets and liabilities:			
Accounts receivable, net	(883)		266
Inventories	30		20
Prepaid expenses and current assets	223		173
Accounts payable	(170)		801
Accrued expenses and other liabilities	(382)		(820)
Net cash used for operating activities	(2,000)		(3,584)
Investing activities:			
Purchases of property and equipment	(62)		(196)
Sales and maturities of marketable securities	—		2,639
Capitalized software development costs			(42)
Net cash provided by (used for) investing activities	(62)		2,401
Effect of exchange rate on changes on cash	(19)		195
Decrease in cash and cash equivalents	(2,081)		(988)
Cash and cash equivalents at beginning of period	2,656		3,564
Cash and cash equivalents at end of period	\$ 575	\$	2,576

See accompanying notes to the condensed consolidated financial statements.

#### IMAGE SENSING SYSTEMS, INC. Notes to Condensed Consolidated Financial Statements (Unaudited) June 30, 2015

### Note A: Basis of Presentation

Image Sensing Systems, Inc. (referred to herein as "we," the "Company," "us" and "our") develops and markets software-based computer enabled detection products for use in traffic, security, police and parking applications. We sell our products primarily to distributors and also receive royalties under a license agreement with a manufacturer/distributor for certain of our products. Our products are used primarily by governmental entities.

The accompanying unaudited condensed consolidated financial statements of the Company have been prepared in accordance with U.S. generally accepted accounting principles (GAAP) for interim financial information and with the instructions to the Quarterly Report on Form 10-Q, which require the Company to make estimates and assumptions that affect amounts reported. Certain information and footnote disclosures normally included in financial statements prepared in accordance with GAAP have been condensed or omitted pursuant to rules and regulations of the Securities and Exchange Commission (the "SEC"). Accordingly, they do not include all of the information and footnotes required by GAAP for complete financial statements. It is the opinion of management that the unaudited condensed consolidated financial statements include all adjustments consisting of normal recurring accruals considered necessary for a fair presentation. All significant intercompany balances and transactions have been eliminated.

Operating results for the three-month and six-month periods ended June 30, 2015 are not necessarily indicative of the results that may be expected for the year ending December 31, 2015. The accompanying condensed consolidated financial statements of the Company should be read in conjunction with the audited consolidated financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2014 as filed with the SEC.

#### Summary of Significant Accounting Policies

The Company believes that of its significant accounting policies, the following are particularly important to the portrayal of the Company's results of operations and financial position and may require the application of a higher level of judgment by the Company's management and, as a result, are subject to an inherent degree of uncertainty.

#### Revenue Recognition

We recognize revenue on a sales arrangement when it is realized or realizable and earned, which occurs when all of the following criteria have been met: persuasive evidence of an arrangement exists; delivery and title transfer have occurred or services have been rendered; the sales price is fixed and determinable; collectability is reasonably assured; and all significant obligations to the customer have been fulfilled.

Certain sales may contain multiple elements for revenue recognition purposes. We consider each deliverable that provides value to the customer on a standalone basis as a separable element. Separable elements in these arrangements may include the hardware, software, installation services, training and support. We initially allocate consideration to each separable element using the relative selling price method. Selling prices are determined by us based on either vendor-specific objective evidence ("VSOE") (the actual selling prices of similar products and services sold on a standalone basis) or, in the absence of VSOE, our best estimate of the selling price. Factors considered by us in determining estimated selling prices for applicable elements generally include overall economic conditions, customer demand, costs incurred by us to provide the deliverable, as well as our historical pricing practices. Under these arrangements, revenue associated with each delivered element is recognized in an amount equal to the lesser of the consideration initially allocated to the delivered element or the amount for which payment is not deemed contingent upon future delivery of other elements in the arrangement. Under arrangements where special acceptance protocols exist, installation services and training may not be considered separable. Under those circumstances, revenue for the entire arrangement is recognized upon the completion of installation, training and fulfillment of any other significant obligations specific to the terms of the arrangement. Arrangements that do not contain any separable elements are typically recognized when the products are shipped and title has transferred to the customer.

Revenue from arrangements for services such as maintenance, repair, consulting and technical support are recognized either as the service is performed or ratably over the defined contractual period for service maintenance contracts.

Econolite Control Products, Inc. (Econolite) is our licensee that sells certain of our products in the United States, Mexico, Canada and the Caribbean. The royalty of approximately 50% of the gross profit on licensed products is recognized when the products are shipped or delivered by Econolite to its customers.

We record provisions against sales revenue for estimated returns and allowances in the period when the related revenue is recorded based on historical sales returns and changes in end user demand.

Revenue is recorded net of taxes collected from customers that are remitted to governmental authorities, with the collected taxes recorded as current liabilities until remitted to the relevant government authority.

#### Inventories

Inventories are primarily electronic components and finished goods and are valued at the lower of cost or market on the first-in, first-out accounting method.

#### Income Taxes

We record a tax provision for the anticipated tax consequences of the reported results of operations. Deferred tax assets and liabilities are measured using the currently enacted tax rates that apply to taxable income in effect for the years in which those deferred tax assets and liabilities are expected to be realized or settled. We record a valuation allowance to reduce deferred tax assets to the amount that is believed more likely than not to be realized. We believe it is more likely than not that forecasted income, including income that may be generated as a result of certain tax planning strategies, together with the tax effects of the deferred tax assets are determined not to be realizable in the future, an adjustment to the valuation allowance would be charged to earnings in the period such determination is made. In addition, the calculation of tax liabilities involves significant judgment in estimating the impact of uncertainties in the application of complex tax laws. Resolution of these uncertainties in a manner inconsistent with management's expectations could have a material impact on our financial condition and operating results. We recognize penalties and interest expense related to unrecognized tax benefits in income tax expense.

#### Intangible Assets

Intangible assets with finite lives are amortized on a straight-line basis over the expected period to be benefited by future cash flows and reviewed for impairment. At both June 30, 2015 and December 31, 2014, we determined there was no impairment of intangible assets. At both June 30, 2015 and December 31, 2014, there were no indefinite-lived intangible assets.

We capitalize certain software development costs related to software to be sold, leased, or otherwise marketed. Capitalized software development costs include purchased materials and services and other costs associated with the development of new products and services. Software development costs are expensed as incurred until technological feasibility has been established, at which time future costs incurred are capitalized until the product is available for general release to the public. A certain amount of judgment and estimation is required to assess when technological feasibility is established, as well as the ongoing assessment of the recoverability of capitalized costs. In evaluating the recoverability of capitalized software costs, the Company compares expected product performance, utilizing forecasted revenue amounts, to the total costs incurred to date and estimates of additional costs to be incurred. If revised forecasted product revenue is less than, and/or revised forecasted costs are greater than, the previously forecasted amounts, the net realizable value may be lower than previously estimated, which could result in the recognition of an impairment charge in the period in which such a determination is made.

#### Note B: Recent Accounting Pronouncements

In May 2014, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2014-09, "Revenue from Contracts with Customers." ASU 2014-09 provides new guidance related to how an entity should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services. In addition, ASU 2014-09 specifies new accounting for costs associated with obtaining or fulfilling contracts with customers and expands the required disclosures related to revenue and cash flows from contracts with customers.

On July 9, 2015, FASB affirmed its proposal to defer the effective date of ASU 2014-09 for all entities by one year. As a result, public business entities, certain not-for-profit entities, and certain employee benefit plans will apply the new revenue standard to annual reporting periods beginning after December 15, 2017. All other entities to apply ASU 2014-09 early, but not before the original effective date for public business entities, certain not-for-profit entities, and certain employee benefit plans (that is, annual periods beginning after December 15, 2016). Entities choosing to implement early will apply the new revenue standard to all interim reporting periods within the year of adoption.

The Company is currently determining its implementation approach and assessing the impact of ASU 2014-09 on the condensed consolidated financial statements.

#### Note C: Fair Value Measurements

The guidance for fair value measurements establishes the authoritative definition of fair value, sets out a framework for measuring fair value and outlines the required disclosures regarding fair value measurements. Fair value is the price that would be received to sell an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants at the measurement date. We use a three-tier fair value hierarchy based upon observable and non-observable inputs as follows:

- Level 1 observable inputs such as quoted prices in active markets;
- Level 2 inputs, other than the quoted prices in active markets, that are observable either directly or indirectly; and
- Level 3 unobservable inputs in which there is little or no market data, which require the reporting entity to develop its own assumptions.

#### Nonfinancial Assets Measured at Fair Value on a Nonrecurring Basis

Our intangible assets and other long-lived assets are nonfinancial assets that were acquired either as part of a business combination, individually or with a group of other assets. These nonfinancial assets were initially, and have historically been, measured and recognized at amounts equal to the fair value determined as of the date of acquisition.

#### Financial Instruments not Measured at Fair Value

Certain of our financial instruments are not measured at fair value and are recorded at carrying amounts approximating fair value, based on their short-term nature or variable interest rate. These financial instruments include cash and cash equivalents, accounts receivable, accounts payable and other current assets and liabilities.

#### Note D: Inventories

Inventories consisted of the following (in thousands):

	June	30,	December 31,			
	2015			2014		
Components	\$	1,742	\$	1,760		
Finished goods		462		474		
	\$	2,204	\$	2,234		



#### Note E: Intangible Assets

Intangible assets consisted of the following (dollars in thousands):

	 June 30, 2015								
	Gross Carrying	Aco	cumulated		Net Carrying	Weighted Average Useful Life			
	Amount	Am	ortization		Value	(in Years)			
Developed technology	\$ 8,121		(6,219)	\$	1,902	2.2			
Trade names	3,267		(2,495)		772	3.0			
Other intangible assets	 1,796		(1,242)		554	1.7			
Total	\$ 13,184	\$	(9,956)	\$	3,228	2.2			

	December 31, 2014							
	Gross Carrying Accumulated Amount Amortization				Net Carrying Value	Weighted Average Useful Life (in Veere)		
	 					(in Years)		
Developed technology	\$ 8,114	\$	(5,666)	\$	2,448	2.6		
Trade names	3,267		(2,367)		900	3.5		
Other intangible assets	1,777		(1,138)		639	2.2		
Total	\$ 13,158	\$	(9,171)	\$	3,987	2.7		

#### Note F: Credit Facilities

In May 2014, the Company entered into a credit agreement and related documents with Alliance Bank providing for a revolving line of credit for the Company. The credit agreement and related documents with Alliance Bank (collectively, the "Alliance Credit Agreement") provide up to a \$5.0 million revolving line of credit. Amounts due under the Alliance Credit Agreement bear interest at a fixed annual rate of 3.95%. Any advances are secured by the Company's inventories, accounts receivable, cash, marketable securities, and equipment. We are subject to certain covenants under the Alliance Credit Agreement. In March 2015, we entered into an agreement with Alliance Bank amending the Alliance Credit Agreement to extend the maturity date from May 12, 2015 to April 1, 2016. At June 30, 2015, we had no borrowings under the Alliance Credit Agreement, and we were in compliance with all financial covenants.

Prior to May 12, 2014, we had a revolving line of credit with Associated Bank, National Association ("Associated Bank") that was initially entered into as of May 1, 2008. We requested, and Associated Bank granted, a termination to this credit facility effective on May 12, 2014 in connection with the revolving line of credit from Alliance Bank described above.

#### Note G: Warranties

We generally provide a standard two-year warranty on product sales. Reserves to honor warranty claims are estimated and recorded at the time of sale based on historical claim information and are analyzed and adjusted periodically based on claim trends.

Warranty liability and related activity consisted of the following (in thousands):

	Six-Month Periods Ended June 30,				
	2015	2	2014		
Beginning balance	\$ 966	\$	934		
Warranty provisions	59		53		
Warranty claims	(103)		(173)		
Adjustments to preexisting warranties	(151)		(81)		
Ending balance	\$ 771	\$	733		

#### Note H: Stock-Based Compensation

We compensate officers, directors and key employees with stock-based compensation under stock option and incentive plans approved by our shareholders and administered under the supervision of our Board of Directors. Stock option awards are granted at exercise prices equal to the closing price of our stock on the day before the date of grant. Generally, options vest proportionally over periods of three to five years from the dates of the grant, beginning one year from the date of grant, and have a contractual term of nine to ten years.

Performance stock options are time based; however, the final number of awards earned and the related compensation expense is adjusted up or down to the extent the performance target is met. The actual number of shares that will ultimately vest ranges from 90% to 100% of the targeted amount if the minimum performance target is achieved. For performance stock awards granted in 2015, the performance target is operating income. We evaluate the likelihood of meeting the performance target at each reporting period and adjust compensation expense, on a cumulative basis, based on the expected achievement of each performance target.

Compensation expense, net of estimated forfeitures, is recognized ratably over the vesting period. Stock-based compensation expense included in general and administrative expense for the three-month periods ended June 30, 2015 and 2014 was \$76,000 and \$99,000, respectively. Stock-based compensation expense included in general and administrative expense for the six-month periods ended June 30, 2015 and 2014 was \$151,000 and \$186,000, respectively. At June 30, 2015, a total of 392,140 shares were available for grant under the Company's stock option and incentive plan.

#### Stock Options

A summary of the option activity for the first six months of 2015 is as follows:

	Number of Shares	Weighted Average Exercise Price per Share	Weighted Average Remaining Contractual Term (in years)	Aggregate Intrinsic Value
Options outstanding at December 31, 2014	354,000	\$ 6.30	7.0	\$ _
Granted	50,000	\$ 2.73	5.0	\$ _
Exercised		\$ 	—	\$ —
Expired	(3,000)	\$ 9.22		\$ —
Forfeited	(54,000)	\$ 5.57	_	\$ 
Options outstanding at June 30, 2015	347,000	\$ 5.88	6.9	\$ 37,000
Options exercisable at June 30, 2015	155,750	\$ 7.23	5.2	\$ _

There were no options exercised during the three-month and six-month periods ended June 30, 2015 and June 30, 2014. As of June 30, 2015, there was \$347,000 of total unrecognized compensation cost related to non-vested stock options. The weighted average period over which the compensation cost is expected to be recognized is 2.3 years.



#### Stock Awards

We issue stock awards as a portion of the annual retainer for each director on a quarterly basis. The stock awards are fully vested at the time of issuance. Compensation expense related to stock awards is determined on the grant date based on the publicly quoted fair market value of our common stock and is charged to earnings on the grant date. During the quarter ended June 30, 2015, there were stock awards issued for 7,860 shares with a weighted-average grant date fair value of \$3.18. For the six months ended June 30, 2015, there were stock awards issued for 17,700 shares with a weighted-average grant date fair value of \$2.82.

#### Note I: Loss per Common Share

Net loss per share is computed by dividing net loss by the daily weighted average number of common shares outstanding during the applicable periods. Diluted net loss per share includes the potentially dilutive effect of common shares subject to outstanding stock options using the treasury stock method. Under the treasury stock method, shares subject to certain outstanding stock options have been excluded from the diluted weighted average shares outstanding calculation because the exercise of those options would lead to a net reduction in common shares outstanding. As a result, stock options to acquire 351,000 and 343,000 weighted common shares have been excluded from the diluted weighted shares outstanding for the three-month periods ended June 30, 2015 and 2014, respectively, and 557,000 and 337,000 weighted common shares have been excluded from the diluted weighted shares outstanding for the six-month periods ended June 30, 2015 and 2014, respectively. The potentially dilutive effect of common shares subject to certain outstanding stock options is determined based on net loss. A reconciliation of these amounts is as follows (dollar amounts in thousands except per share data):

	Three-Month Periods Ended June 30,					ed		
		2015	2014		2015			2014
Numerator:								
Net loss	\$	(742)	\$	(1,533)	\$	(1,909)	\$	(5,290)
Denominator:								
Weighted average common shares outstanding		5,008		4,980		5,003		4,978
Dilutive potential common shares		_		—				
Shares used in diluted net loss per common share calculations		5,008		4,980		5,003		4,978
Basic net loss per common share	\$	(0.15)	\$	(0.31)	\$	(0.38)	\$	(1.06)
Diluted net loss per common share	\$	(0.15)	\$	(0.31)	\$	(0.38)	\$	(1.06)

#### Note J: Segment Information

The Company's Interim Chief Executive Officer and management regularly review financial information for the Company's three discrete operating segments. Based on similarities in the economic characteristics, nature of products and services, production processes, type or class of customer served, method of distribution and regulatory environments, the operating segments have been aggregated for financial statement purposes and categorized into three reportable segments: Intersection, Highway and License Plate Recognition ("LPR"). Autoscope® video is our machine-vision product line, and revenue consists of royalties (all of which are received from Econolite), as well as a portion of international product sales. Video products are normally sold in the Intersection segment. RTMS® is our radar product line, and revenue consists of international and North American product sales as well as a portion of royalties (all of which are received from Econolite). Radar products are normally sold in the Highway segment. Until July 9, 2015, Autoscope® license plate recognition was our LPR product line. All segment revenues are derived from external customers. As described in Note M to these Notes to the Condensed Consolidated Financial Statements, effective on July 9, 2015, we sold our LPR business.

Operating expenses and total assets are not allocated to the segments for internal reporting purposes. Due to the changes in how we manage our business, we may reevaluate our segment definitions in the future.



763

3,228

538

5,065

782

5,763

#### Table of Contents

The following tables set forth selected unaudited financial information for each of our reportable segments (in thousands):

				Three Months	Ended June 30,				
	Inters	ection	Hig	hway	LP	R	Total		
	2015	2014	2015	2014	2015	2014	2015	2014	
Revenue	\$ 3,189	\$ 3,625	\$ 1,218	\$ 1,094	\$ 774	\$ 1,219	\$ 5,181	\$ 5,938	
Gross profit	2,768	3,242	783	488	653	533	4,204	4,263	
Amortization of intangible assets			122	121	260	272	382	393	
Intangible assets	—	—	210	698	3,018	5,065	3,228	5,763	
				Six Months E	Ended June 30,				
	Inters	ection	Hig	hway	LP	R	To	otal	
	2015	2014	2015	2014	2015	2014	2015	2014	
Revenue	\$ 5,350	\$ 5,808	\$ 2,284	\$ 2,169	\$ 1,886	\$ 2,283	\$ 9,520	\$ 10,260	
Gross profit	4,828	5,347	1,363	1,166	1,333	867	7,524	7,380	

#### Note K: Restructuring

Intangible assets

Amortization of intangible assets

In the fourth quarter of 2014, the Company implemented restructuring plans to close our offices in Asia. Because of these actions, restructuring charges of approximately \$119,000 were recorded in the first six months of 2015 related to employee terminations.

244

210

244

698

519

3,018

The following table shows the restructuring activity for 2015 (in thousands):

	nination nefits	and C	ty Costs Contract lination	Total		
Balance at January 1, 2015	\$ 190	\$	26	\$	216	
Charges	119				119	
Payments/settlements	(295)		(8)		(303)	
Balance at March 31, 2015	\$ 14	\$	18	\$	32	
Payments/settlements	 (14)		(18)		(32)	
Balance at June 30, 2015	\$ 	\$		\$		

In the first quarter of 2014, the Company implemented restructuring plans to improve our financial performance in Europe. These plans included the closure of our office in Poland. Because of these actions, restructuring charges of approximately \$460,000 were recorded related primarily to facilities and employee terminations.

The following table shows the restructuring activity for 2014 (in thousands):

		nination nefits	and	lity Costs Contract mination	 Total
Balance at January 1, 2014		\$ _	\$	_	\$ _
Charges		60		400	460
Payments/settlements		(45)		(177)	(222)
Balance at March 31, 2014		\$ 15	\$	223	\$ 238
Payments/settlements		(15)		(223)	(238)
Balance at June 30, 2014		\$ 	\$		\$ 
	13				

#### Note L: Commitments and Contingencies

#### Litigation

We are involved from time to time in various legal proceedings arising in the ordinary course of our business, including primarily commercial, product liability, employment and intellectual property claims. In accordance with GAAP, we record a liability in our Consolidated Financial Statements with respect to any of these matters when it is both probable that a liability has been incurred and the amount of the liability can be reasonably estimated. With respect to any currently pending legal proceedings, we have not established an estimated range of reasonably possible additional losses either because we believe that we have valid defenses to claims asserted against us or the proceeding has not advanced to a stage of discovery that would enable us to establish an estimate. We currently do not expect the outcome of these matters to have a material effect on our consolidated results of operations, financial position or cash flows. Litigation, however, is inherently unpredictable, and it is possible that the ultimate outcome of one or more claims asserted against us could adversely impact our results of operations, financial position or cash flows. We expense legal costs as incurred.

#### Note M: Subsequent Events

On July 9, 2015, we announced the sale of our LPR business to TagMaster A.B. ("TagMaster") located in Stockholm, Sweden, for the purchase price of \$4.2 million in cash. As of July 9, 2015, the LPR business, including all products and solutions, transitioned to TagMaster. We have included a pro forma statement of operations as if the disposition occurred on April 1, 2015.

The following unaudited pro forma statement of operations gives effect to the disposition of the Image Sensing Systems LPR business segment and any associated transaction costs incurred in the second quarter as if the disposition occurred on April 1, 2015. The pro forma adjustments, described in the related notes, are based on the best available information and certain assumptions that ISS management believes are reasonable.

The unaudited consolidated pro forma statement of operations is presented for illustrative purposes only and does not purport to be indicative of the operating results that would have occurred if the transaction described above had occurred as presented in such statement. For example, this financial information does not reflect any potential earnings or other impacts from the use of the proceeds from the disposition or cost reductions previously allocated corporate costs and potential subsequent restructuring charges.

## Image Sensing Systems, Inc. Unaudited Pro Forma Consolidated Statement of Operations for the three months Ended June 30, 2015 (in thousands, except per share data)

	S Cor	ge Sensing systems isolidated istorical	Adjus	o Forma tments LPR sposition	S	age Sensing ystems Pro Forma onsolidated
Consolidated Statement of Operations Data:						
Revenue:						
Product sales	\$	2,572	\$	774(a)	\$	1,798
Royalties		2,609				2,609
		5,181		774		4,407
Cost of revenue:						
Product sales		977		121(a)		856
		977		121		856
Gross profit		4,204		653		3,551
Operating expenses:						
Selling, marketing and product support		1,322		466(b)		856
General and administrative		1,390		410(b)		980
Research and development		1,123		164(b)		959
Amortization of intangible assets		382		259(b)		123
LPR business unit sale costs		751		751(c)		125
		4,968		2,050		2,918
Income (Loss) from operations		(764)		(1,397)		633
income (Loss) from operations		(704)		(1,597)		055
Other expense, net		30				30
Loss before income taxes		(734)		(1,397)		663
Income tax expense		8		_		8
Net income (loss)	\$	(742)	\$	(1,397)	\$	655
Net loss per share:						
Basic	¢	(0, 15)	¢	(0.20)	¢	0.12
	\$	(0.15)	\$	(0.28)	\$	0.13
Diluted	<u>\$</u>	(0.15)	<u>\$</u>	(0.28)	\$	0.13
Weighted average number of common shares outstanding:						
Basic		5,008		5,008		5,008
Diluted		5,008		5,008		5,008
Diruwa		5,000		5,000		5,008

Notes:

Represents the decrease of LPR revenue and cost of sales as if the divestiture occurred on April 1, 2015. Represents the pro forma decrease of operating cost as if the LPR divestiture occurred on April 1, 2015. (a) (b)

Represents the LPR business unit sale costs. (c)

## Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

#### Overview

General. We provide software based computer enabled detection ("CED") products and solutions that use advanced signal processing software algorithms to detect and monitor objects in a designated field of view. Our technology analyzes signals from a sophisticated sensor and passes the information along to management systems, controllers or directly to users. Our core products, the Autoscope® Video Vehicle Detection System, RTMS® Detection System and, until July 9, 2015, Autoscope® License Plate Recognition ("LPR") System, operate using our proprietary application software in conjunction with video cameras or radar and commonly available electronic components. Our systems are used by traffic managers primarily to improve the flow of vehicle traffic and to enhance safety at intersections, main thoroughfares, freeways and tunnels and by parking and toll managers and law enforcement officials to read license plates for various safety, security, access and enforcement LPR applications.

Autoscope® video systems are sold to distributors and end users of traffic management products in the United States, Mexico, Canada and the Caribbean by Econolite Control Products, Inc. ("Econolite"), our exclusive licensee in these regions. Until July 9, 2015, we sold LPR systems to distributors and end users in the United States, Canada and Mexico. We sell all of our systems to distributors and end users in Europe and Asia through our subsidiaries. The majority of our sales are to end users that are funded by government agencies responsible for traffic management or traffic law enforcement.

**RTMS®** Radar Business Model Change. From July 24, 2012 until July 14, 2014, our RTMS systems also were sold by Econolite in the United States, Mexico, Canada and the Caribbean under the same arrangement as Econolite sold our Autoscope® video systems. Effective July 14, 2014, the marketing, manufacturing and distribution of the RTMS® radar product line in these regions transitioned from Econolite to the Company.

*Sale of LPR Business.* On July 9, 2015, we closed the sale of our LPR business to TagMaster A.B. ("TagMaster") for the purchase price of \$4.2 million in cash under the Share and Asset Sale and Purchase Agreement dated July 9, 2015 (the "SAPA") by and among Image Sensing Systems, Inc.; Image Sensing Systems EMEA Limited, our wholly-owned subsidiary ("ISS EMEA"); and TagMaster. We incurred \$751,000 and \$857,000 in costs related to the sale in the three month and six month periods ended June 30, 2015, respectively, which were included in operating costs. We described the sale of the LPR business unit in our Current Reports on Form 8-K filed with the Securities and Exchange Commission on July 10, 2015 and July 15, 2015.

A copy of the SAPA is filed as Exhibit 2.1 to this Quarterly Report on Form 10-Q. The representations, warranties and covenants of each of Image Sensing Systems, Inc., ISS EMEA and TagMaster contained in the SAPA have been made solely for the benefit of the parties to the SAPA. In addition, such representations, warranties and covenants have been made only for purposes of the SAPA; may have been qualified by confidential disclosures made by the parties in connection with the SAPA; are subject to materiality qualifications contained in the SAPA that may differ from what may be viewed as material by investors; were made only as of the date of the SAPA or such other date as is specified in the SAPA; and have been included in the SAPA for the purpose of allocating risk between the contracting parties rather than establishing matters as facts. Accordingly, the SAPA is included as an exhibit to this Quarterly Report on Form 10-Q only to provide investors with information regarding its terms and not to provide investors with any other factual information regarding the parties to the SAPA or their respective businesses. Investors should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of the parties to the SAPA or any of their respective subsidiaries. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the SAPA, which subsequent information may or may not be fully reflected in the public disclosures by the parties or their subsidiaries. The SAPA should not be read alone but should instead be read in conjunction with any other information regarding the parties on their subsidiaries. The SAPA should not be read alone but should instead be read in conjunction with any other information regarding the parties or their subsidiaries. The SAPA should not be read alone but should instead be read in conjunction with any other information regarding the parties that



#### Trends and Challenges in Our Business

We believe the expected growth in our business can be attributed primarily to the following global trends:

- worsening traffic caused by increased numbers of vehicles in metropolitan areas without corresponding expansions of road infrastructure and the need to automate safety, security and access applications for automobiles and trucks, which has increased demand for our products;
- advances in information technology, which have made our products easier to market and implement;
- the continued funding allocations for centralized traffic management services and automated enforcement schemes, which have increased the ability
  of our primary end users to implement our products; and
- general increases in the cost-effectiveness of electronics, which make our products more affordable for end users.

We believe our continued growth primarily depends upon:

- continued adoption and governmental funding of intelligent transportation systems ("ITS") and other automated applications for traffic control, safety and enforcement in developed countries;
- a propensity by traffic engineers to implement lower cost technology-based solutions rather than civil engineering solutions such as widening roadways;
- countries in the developing world adopting above-ground detection technology, such as video or radar, instead of in-pavement loop technology to manage traffic;
- the use of CED to provide solutions to security/surveillance and environmental issues associated with increasing automobile use in metropolitan areas; and
- our ability to develop new products, such as hybrid CED devices incorporating, for example, radar and video technologies, that provide increasingly
  accurate information and enhance the end users' ability to cost-effectively manage traffic, security/surveillance and environmental issues.

Because the majority of our end users are governmental entities, we are faced with challenges related to potential delays in purchase decisions by those entities and changes in budgetary constraints. These contingencies could result in significant fluctuations in our revenue between periods. The ongoing economic environment in Europe and the United States is further adding to the unpredictability of purchase decisions, creating more delays than usual and decreasing governmental budgets, and it is likely to continue to affect our revenue.

#### Key Financial Terms and Metrics

*Revenue*. We derive revenue from two sources: (1) royalties received from Econolite for sales of the Autoscope® video and RTMS® radar (from July 24, 2012 until July 14, 2014) systems in the United States, Mexico, Canada and the Caribbean and (2) revenue received from the direct sales of our RTMS® radar (before July 24, 2012 and after July 14, 2014) and LPR systems in the United States, Mexico, Canada and the Caribbean (until July 9, 2015) and all of our systems in Europe and Asia. On July 14, 2014, we announced the transfer of North American marketing and manufacturing of the RTMS® radar product line from Econolite to Image Sensing Systems, ending our royalty agreement with Econolite for radar sales. Autoscope® video royalties are calculated using a profit sharing model where the gross profits on sales of product made through Econolite are shared equally with Econolite. This royalty arrangement has the benefit of decreasing our cost of revenues and our selling, marketing and product support expenses because these costs and expenses are borne primarily by Econolite. Although this royalty model has a positive impact on our gross margin, it also negatively impacts our total revenue, which would be higher if all the sales made by Econolite were made directly by us. The royalty arrangement is exclusive under a long-term agreement.

*Cost of Revenue*. There is no cost of revenue related to royalties, as virtually all manufacturing, warranty and related costs are incurred by Econolite. Cost of revenue related to product sales consists primarily of the amount charged by our third party contractors to manufacture hardware platforms, which is influenced mainly by the cost of electronic components. The cost of revenue also includes logistics costs, estimated expenses for product warranties, restructuring costs and inventory reserves. The key metric that we follow is achieving certain gross margin percentages on product sales by geographic region and to a lesser extent by product line.



19 of 28

#### Table of Contents

Operating Expenses. Our operating expenses fall into three categories: (1) selling, marketing and product support; (2) general and administrative; and (3) research and development. Selling, marketing and product support expenses consist of various costs related to sales and support of our products, including salaries, benefits and commissions paid to our personnel; commissions paid to third parties; travel, trade show and advertising costs; second-tier technical support for Econolite; and general product support, where applicable. General and administrative expenses consist of certain corporate and administrative functions that support the development and sales of our products and provide an infrastructure to support future growth. These expenses include management, supervisory and staff salaries and benefits, legal and auditing fees, travel, rent and costs associated with being a public company, such as board of director fees, listing fees and annual reporting expenses. Research and development expenses consist mainly of salaries and benefits for our engineers and third party costs for consulting and prototyping. We measure all operating expenses against our annually approved budget, which is developed with achieving a certain operating margin as a key focus. Also included in operating expenses are restructuring costs and non-cash expense for intangible asset amortization.

Non-GAAP Operating Measure. We provide certain non-GAAP financial information as supplemental information to financial measures calculated and presented in accordance with GAAP (Generally Accepted Accounting Principles in the United States). This non-GAAP information excludes the impact of amortizing intangible assets and depreciation and may exclude other non-recurring items. Management believes that this presentation facilitates the comparison of our current operating results to historical operating results. Management uses this non-GAAP information to evaluate short-term and long-term operating trends in our core operations. Non-GAAP information is not prepared in accordance with GAAP and should not be considered a substitute for or an alternative to GAAP financial measures and may not be computed the same as similarly titled measures used by other companies.

Reconciliations of GAAP net income (loss) to non-GAAP net income and loss are as follows (in thousands):

	 Three-Month June	nded		ded			
	 2015		2014		2015		2014
Loss from operations	\$ (764)	\$	(1,558)	\$	(1,914)	\$	(5,314)
Adjustments to reconcile to non-GAAP net loss							
Amortization of intangible assets	382		393		763		782
Depreciation	85		145		173		298
Sale of business unit	751				857		
Restructuring charges					119		460
Investigation matter			36				152
Non-GAAP net income (loss)	\$ 454	\$	(984)	\$	(2)	\$	(3,622)

Seasonality. Our quarterly revenues and operating results have varied significantly in the past due to the seasonality of our business. Our first quarter generally is the weakest due to weather conditions that make roadway construction more difficult in parts of North America. Europe and northern Asia. We expect such seasonality to continue for the foreseeable future. Additionally, our international revenues regularly contain individually significant sales. This can result in significant variations of revenue between periods. Accordingly, we believe that quarter-to-quarter comparisons of our financial results should not be relied upon as an indication of our future performance. No assurance can be given that we will be able to achieve or maintain profitability on a quarterly or annual basis in the future

Segments. As of June 30, 2015, we operated in three reportable segments: Intersection, Highway and LPR. Autoscope® video is our machine-vision product line, and revenue consists of royalties (all of which are received from Econolite), as well as a portion of international product sales. Video products are normally sold in the Intersection segment. The RTMS® is our radar product line, and revenue consists of sales to external customers. Radar products are normally sold in the Highway segment. Until July 9, 2015, Autoscope® license plate recognition was our LPR product line. All segment revenues are derived from external customers. As described elsewhere in this Quarterly Report on 10-Q, effective on July 9, 2015, we sold our LPR business. As a result of business model changes and modifications in how we manage our business, we may reevaluate our segment definitions in the future.

The following tables set forth selected unaudited financial information for each of our reportable segments (in thousands):

	Three Months Ended June 30,															
		Intersection Highway						LPR					Total			
		2015		2014	_	2015		2014	_	2015	_	2014	_	2015	_	2014
Revenue	\$	3,189	\$	3,625	\$	1,218	\$	1,094	\$	774	\$	1,219	\$	5,181	\$	5,938
Gross profit		2,768		3,242		783		488		653		533		4,204		4,263
Amortization of intangible		,		,		100		101		2(0		272		le la		
assets						122		121		260		272		382		393
Intangible assets						210		698		3,018		5,065		3,228		5,763
								Six Months E	ndec	l June 30,						
		Inters	sectior	1		High	iway		LPR				Total			
		2015		2014	_	2015		2014	_	2015	_	2014	_	2015		2014
Revenue	\$	5,350	\$	5,808	\$	2,284	\$	2,169	\$	1,886	\$	2,283	\$	9,520	\$	10,260
Gross profit		4,828		5,347		1,363		1,166		1,333		867		7,524		7,380
Amortization of intangible		,		,		í.								le la		
assets						244		244		519		538		763		782
Intangible assets						210		698		3,018		5,065		3,228		5,763
						2	0									

Period

### Table of Contents

#### **Results of Operations**

The following table sets forth, for the periods indicated, certain statements of operations data as a percent of total revenue and gross profit on product sales and royalties as a percentage of product sales and royalties, respectively.

	Three-Month Periods June 30,	Ended	Quarter Over Quarter
	2015	2014	Change
Product sales	49.6%	44.6%	(3.0)%
Royalties	50.4	55.4	(20.6)
Total revenue	100.0	100.0	(12.7)
Gross profit - product sales	62.0	36.8	63.4
Gross profit - royalties	100.0	100.0	(20.6)
Selling, marketing and product support	25.5	41.6	(46.5)
General and administrative	26.8	25.2	(7.2)
Research and development	21.7	24.0	(21.1)
Amortization of intangible assets	7.4	6.6	(2.8)
Sale of business unit	14.5		_
Investigation matter	—	0.6	(100.0)
Loss from operations	(14.7)	(26.2)	(51.0)
Income tax expense (benefit)	0.2	(0.2)	(180.0)
Net loss	(14.3)	(25.8)	(51.6)

	Six-Month Periods	Ended	Over Period
	2015	2014	Change
Product sales	51.5%	44.3%	7.7%
Royalties	48.5	55.7	(19.1)
Total revenue	100.0	100.0	(7.2)
Gross profit - product sales	59.3	36.7	74.0
Gross profit - royalties	100.0	100.0	(19.1)
Selling, marketing and product support	28.8	50.6	(47.3)
General and administrative	29.0	28.0	(3.8)
Research and development	23.1	31.6	(32.1)
Amortization of intangible assets	8.0	7.6	(2.4)
Sale of business unit	9.0	—	—
Restructuring	1.3	4.5	(74.1)
Investigation matter		1.5	(100.0)
Loss from operations	(20.1)	(51.8)	(64.0)
Income tax expense (benefit)	0.3	(0.1)	(340.0)

Total revenue decreased to \$5.2 million in the three month period ended June 30, 2015 from \$5.9 million in the same period in 2014, a decrease of 12.7%, and to \$9.5 million in the first half of 2015 from \$10.3 million in the first half of 2014, a decrease of 7.2%. Royalty income decreased to \$2.6 million in the second quarter of 2015 from \$3.3 million in the second quarter of 2014, a decrease of 20.6%, and decreased to \$4.6 million in the first half of 2015 from \$5.7 million in the first half of 2014, a decrease of 19.1%. The decrease in royalties was the result of a decrease in Autoscope® video system sales under the Econolite agreement and the transition of RTMS® radar product sales from Econolite to the Company. Autoscope® video royalties were lower in the three months ended June 30, 2015 compared to the three months ended June 30, 2014 as a result of lower unit volume.



Product sales decreased to \$2.6 million in the second quarter of 2015 from \$2.7 million in the second quarter of 2014, a decrease of 3.0%, and increased to \$4.9 million in the first half of 2015 from \$4.5 million in the same period in 2014, an increase of 7.7%. The increase in product sales was mainly due to higher sales volume of RTMS® radar sales in North America due to the previously discussed radar transition, partially offset by lower sales volume in Asia.

Revenue for the Intersection segment decreased to \$3.2 million in the three-month period ended June 30, 2015 from \$3.6 million in the three-month period ended June 30, 2014, a decrease of 12.0%. Revenue for the Intersection segment decreased to \$5.3 million in the first six months of 2015 from \$5.8 million in the first six months of 2014, a decrease of 7.9%. The decrease in revenue for the Intersection segment was mainly due to lower sales volume.

Revenue for the Highway segment increased to \$1.2 million in the three-month period ended June 30, 2015 from \$1.1 million in the three-month period ended June 30, 2014, an increase of 11.3%. Revenue for the Highway segment increased to \$2.3 million in the first six months of 2015 from \$2.2 million in the first six months of 2014, an increase of 5.3%. The nominal increase in revenue for the Highway segment was due mainly to the transition of North American Highway sales from Econolite back to Image Sensing Systems in the third quarter of 2014. This transition resulted in the direct sale of product sales to customers instead of a royalty from Econolite.

Revenue for the LPR segment decreased to \$774,000 in the period ended June 30, 2015 from \$1.2 million in the same period for 2014, a decrease of 36.5%, and decreased to \$1.9 million in the first half of 2015 from \$2.3 million in the first half of 2014, a decrease of 17.3%. The decrease in revenue for the LPR segment is due to lower sales volume.

Gross profit for product sales increased to 62.0% in the three months ended June 30, 2015 from 36.8% in the three months ended June 30, 2014 and increased to 59.3% in the first six months of 2015 from 36.7% in the first six months of 2014. Gross profit for the LPR product line has historically been lower than gross profit for the Intersection and Highway product lines and therefore the mix of the product lines sold in any given period can result in varying gross profit. Generally, lower sales volumes of Highway or LPR products will reduce gross profit because of fixed manufacturing costs for these products. Additionally, the geographic sales mix of our product sales can influence margins, as product sold in some jurisdictions have lower margins. We anticipate that gross profit for our product sales will be higher in 2015 as compared to 2014, while we expect royalty gross profit will be 100% in 2015.

Selling, marketing and product support expense decreased to \$1.3 million or 25.5% of total revenue in the second guarter of 2015 from \$2.5 million or 41.6% of total revenue in the second quarter of 2014, and decreased to \$2.8 million or 28.8% of total revenue in the first six months of 2015 from \$5.2 million or 50.6% of total revenue in the first six months of 2014. Our selling, marketing and product support expense decreased mainly due to cost reductions, which were partially offset by royalties due to Econolite for orders that were received by Econolite but fulfilled by Image Sensing Systems. We anticipate that annual selling, marketing and product support expense will decrease in dollar amount in 2015 as compared to 2014.

General and administrative expense decreased to \$1.4 million or 26.8% of total revenue for the three months ended June 30, 2015 from \$1.5 million or 25.2% of total revenue for the three months ended June 30, 2014, and decreased to \$2.8 million or 29.0% of total revenue for the first half of 2015 from \$2.9 million or 28.0% of total revenue in the first half of 2014. Overall we anticipate that annual general and administrative expenses will decrease in 2015 as compared to 2014.

Research and development expense decreased to \$1.1 million or 21.7% of total revenue in the three-month period ended June 30, 2015 from \$1.5 million or 24.0% of total revenue in the three-month period ended June 30, 2014, and decreased to \$2.2 million or 23.1% of total revenue in the first six months of 2015 from \$3.2 million or 31.6% of total revenue in the first six months of 2014. The decrease is primarily driven by investment as it relates to current and anticipated research and development projects in 2015. We anticipate that research and development expense will decrease in 2015 compared to 2014.

In the first quarter of 2014, the Company implemented restructuring plans to improve our financial performance in Europe. These plans included the closure of our office in Poland. Because of these actions, restructuring charges of approximately \$460,000 were recorded related primarily to the closure of facilities and employee terminations. In the fourth quarter of 2014, the Company implemented restructuring plans to close our offices in Asia. Because of these actions, additional restructuring charges of approximately \$119,000 were recorded in the first quarter of 2015. We expect costs related to our Asia office closures to be de minimis in future periods.

Amortization of intangibles was \$382,000 in the second quarter of 2015 and \$763,000 in the first half of 2015 compared to \$393,000 in the second quarter of 2014 and \$782,000 in the first half of 2014. This reflects the amortization of intangible assets acquired in acquisitions. Assuming there are no changes to our intangible assets, we anticipate amortization expense will be approximately \$1.5 million in 2015.

In the three-month and six-month periods ended June 30, 2015, we incurred \$751,000 and \$857,000, respectively, in costs related to the sale of our LPR business unit. We incurred no such costs in the first quarter or first half of 2014.

We recorded income tax expense of \$8,000, or 1.1% of pretax loss, for the three months ended June 30, 2015, and income tax expense of \$24,000, or 1.3% of pretax loss, for the six months ended June 30, 2015. We recorded an income tax benefit of \$10,000 for the three and six months ended June 30, 2014. Certain jurisdictions have net operating loss carry forwards.

#### Liquidity and Capital Resources

At June 30, 2015, we had \$575,000 in cash and cash equivalents compared to \$2.7 million in cash and cash equivalents at December 31, 2014. On July 9, 2015, we closed on the sale of our LPR business unit for the purchase price of \$4.2 million in cash. We plan to use the cash from the sale to strengthen our balance sheet and invest in our Autoscope video detection and RTMS® radar detection products and solutions. Our investment objectives are to preserve principal, maintain liquidity, and achieve the best available return consistent with our primary objectives of safety and liquidity.

Net cash used in operating activities was \$2.0 million in the first six months of 2015 compared to cash used in operating activities of \$3.6 million in the same period in 2014. The primary reason for the decrease in cash was the loss for the first six months of 2015, the timing of the collection of outstanding receivables, as well as timing of the payment of outstanding payables, partially offset by the conversion of inventory. We anticipate that average receivable collection days in 2015 will improve from 2014 but that the improvement will not have a material impact on our liquidity.

Net cash used in investing activities was \$62,000 for the first half of 2015 compared to cash provided by investing activities of \$2.4 million in the same period in 2014. Our planned additions of property and equipment are discretionary, and we do not expect them to exceed historical levels in 2015.

In May 2014, the Company entered into a credit agreement and related documents with Alliance Bank providing for a revolving line of credit for the Company. The credit agreement and related documents with Alliance Bank (collectively, the "Alliance Credit Agreement") provide up to a \$5.0 million revolving line of credit. Amounts due under the Alliance Credit Agreement bear interest at a fixed annual rate of 3.95%. Any advances are secured by the Company's inventories, accounts receivable, cash, marketable securities, and equipment. We are subject to certain covenants under the Alliance Credit Agreement. In March 2015, we entered into an agreement with Alliance Bank amending the Alliance Credit Agreement to extend the maturity date from May 12, 2015 to April 1, 2016. At June 30, 2015, we had no borrowings under the Alliance Credit Agreement, and we were in compliance with all financial covenants.

Prior to May 12, 2014, we had a revolving line of credit with Associated Bank, National Association ("Associated Bank") that was initially entered into as of May 1, 2008. We requested, and Associated Bank granted, a termination to this credit facility effective on May 12, 2014 in connection with the revolving line of credit from Alliance Bank described above.

We believe that cash and cash equivalents on hand at June 30, 2015, the cash received from the sale of our LPR business unit on July 9, 2015, along with the availability of funds under our revolving line of credit and cash provided by operating activities, will satisfy our projected working capital needs, investing activities, and other cash requirements for the foreseeable future.

#### Off-Balance Sheet Arrangements

We do not participate in transactions or have relationships or other arrangements with an unconsolidated entity, including special purpose and similar entities, or other off-balance sheet arrangements.

#### **Critical Accounting Policies**

Our significant accounting policies are described in Note 1 to the Consolidated Financial Statements in our Annual Report on Form 10-K for the year ended December 31, 2014. The accounting policies used in preparing our interim Condensed Consolidated Financial Statements as of and for the three and six months ended June 30, 2015 set forth elsewhere in this Quarterly Report on Form 10-Q are the same as those described in our Annual Report on Form 10-K.

#### Cautionary Statement:

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange of 1934, as amended. Forward-looking statements represent our expectations or beliefs concerning future events and can be identified by the use of forward-looking words such as "expects," "believes," "may," "will," "should," "intends," "plans," "estimates," or "anticipates" or other comparable terminology. Forward-looking statements are subject to risks and uncertainties that may cause our actual results to differ materially from the results described in the forward-looking statements. Factors that might cause such differences include, but are not limited to:

- our historical dependence on a single product for most of our revenue;
- budget constraints by governmental entities that purchase our products, including constraints caused by declining tax revenue;
- the continuing ability of Econolite to pay royalties owed;
- the mix of and margin on the products we sell;
- our dependence on third parties for manufacturing and marketing our products;
- our dependence on single-source suppliers to meet manufacturing needs;
- our failure to secure adequate protection for our intellectual property rights;
- our inability to develop new applications and product enhancements;
- the potential disruptive effect on the markets we serve of new and emerging technologies and applications, including vehicle to vehicle communications;
- unanticipated delays, costs and expenses inherent in the development and marketing of new products;
- our inability to respond to low-cost local competitors in Asia and elsewhere;
- our inability to properly manage any growth in revenue and/or production requirements;
- the influence over our voting stock by affiliates;
- our inability to hire and retain key scientific and technical personnel;
- the effects of legal matters in which we may become involved;
- our inability to achieve and maintain effective internal controls;
- our inability to successfully integrate acquisitions;
- political and economic instability, including continuing volatility in the economic environment of the European Union;
- our inability to comply with international regulatory restrictions over hazardous substances and electronic waste; and
- conditions beyond our control such as war, terrorist attacks, health epidemics and economic recession.

We caution that the forward-looking statements made in this report or in other announcements made by us are further qualified by the risk factors set forth in Item 1A. to our Annual Report on Form 10-K for the fiscal year ended December 31, 2014.

## Item 3. Quantitative and Qualitative Disclosures About Market Risk

Our foreign sales and results of operations are subject to the impact of foreign currency fluctuations. From time to time, we enter into currency hedges to attempt to lower our exposure to translation gains and losses as well as to limit the impact of foreign currency translation upon the consolidation of our foreign subsidiaries. A 10% adverse change in foreign currency rates, if we have not properly hedged, could have a material effect on our results of operations or financial position.

#### Item 4. Controls and Procedures

#### Evaluation of Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our Interim Chief Executive Officer and Chief Financial Officer, we evaluated the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rule 13a-15(e) under the Securities Exchange Act of 1934 (the Exchange Act)). Based upon that evaluation, our Interim Chief Executive Officer and Chief Financial Officer concluded that, as of the end of the period covered by this report, our disclosure controls and procedures were effective.

#### Changes in Internal Control Over Financial Reporting

During the fiscal quarter covered by this Quarterly Report on Form 10-Q, there has been no change in our internal control over financial reporting (as defined in Rule 13a-15(f) under the Exchange Act) that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.



#### **PART II. OTHER INFORMATION**

Item 1. Legal Proceedings

None.

Item 1A. Risk Factors

Some of the risk factors to which we and our business are subject are described in the section entitled "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2014. The risks and uncertainties described in our Annual Report are not the only risks we face. Additional risks and uncertainties not presently known to us or that our management currently deems immaterial also may impair our business operations. If any of the risks described were to occur, our business, financial condition, operating results and cash flows could be materially adversely affected.

Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds
None.	
Item 3.	Defaults Upon Senior Securities
None.	
Item 4.	Mine Safety Disclosures
None.	
Item 5.	Other Information
None.	
Item 6.	Exhibits

The following exhibits are filed as part of this Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2015:

Exhibit Number	Description
2.1	Share and Asset Sale and Purchase Agreement dated as of July 9, 2015 among Image Sensing Systems, Inc., Image Sensing Systems EMEA Limited and TagMaster AB ("SAPA"). (Pursuant to Item 601(b)(2) of Regulation S-K under the Securities Act of 1933 and the Securities Exchange Act of 1934, certain schedules to the SAPA have not been filed, and the SAPA briefly identifies the contents of these schedules. Image Sensing Systems, Inc. hereby agrees to furnish supplementally a copy of any omitted schedules to the Securities and Exchange Commission upon its request.)*
31.1	Certification of Interim Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (filed herewith).
31.2	Certification of Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (filed herewith).
32.1	Certification of Interim Chief Executive Officer and Chief Financial Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (filed herewith).
101	The following financial information from the Quarterly Report on Form 10-Q for the quarter ended June 30, 2015, formatted in XBRL (Extensible Business Reporting Language), (i) the Condensed Consolidated Balance Sheets, (ii) the Condensed Consolidated Statements of Operations, (iii) the Condensed Consolidated Statements of Cash Flows, and (iv) the Notes to Condensed Consolidated Financial Statements (filed herewith).

<sup>\*</sup> Portions of this exhibit are treated as confidential pursuant to a request for confidential treatment filed by Image Sensing Systems, Inc. with the Securities and Exchange Commission.

#### SIGNATURES

In accordance with the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Image Sensing Systems, Inc.

Dated: August 11, 2015

By: /s/ Dale E. Parker

Dale E. Parker Interim President, Interim Chief Executive Officer and Chief Financial Officer (Interim Principal Executive Officer, Principal Financial Officer and Principal Accounting Officer)

#### EXHIBIT INDEX

## Exhibit No. Description

- 2.1 Share and Asset Sale and Purchase Agreement dated as of July 9, 2015 among Image Sensing Systems, Inc., Image Sensing Systems EMEA Limited and TagMaster AB ("SAPA"). (Pursuant to Item 601(b)(2) of Regulation S-K under the Securities Act of 1933 and the Securities Exchange Act of 1934, certain schedules to the SAPA have not been filed, and the SAPA briefly identifies the contents of these schedules. Image Sensing Systems, Inc. hereby agrees to furnish supplementally a copy of any omitted schedules to the Securities and Exchange Commission upon its request.)\*
- 31.1 Certification of Interim Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (filed herewith).
- 31.2 Certification of Interim Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (filed herewith).
- 32.1 Certification of Chief Executive Officer and Chief Financial Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (filed herewith).
- 101 The following financial information from the Quarterly Report on Form 10-Q for the quarter ended June 30, 2015, formatted in XBRL (Extensible Business Reporting Language), (i) the Condensed Consolidated Balance Sheets, (ii) the Condensed Consolidated Statements of Operations, (iii) the Condensed Consolidated Statements of Cash Flows, and (iv) the Notes to Condensed Consolidated Financial Statements (filed herewith).

<sup>\*</sup> Portions of this exhibit are treated as confidential pursuant to a request for confidential treatment filed by Image Sensing Systems, Inc. with the Securities and Exchange Commission.

EX-2.1 152795 - FINAL

## Exhibit 2.1

## DATED 9 July 2015

## TAGMASTER AKTIEBOLAG

#### as Purchaser

#### -and-

#### IMAGE SENSING SYSTEMS EMEA LIMITED

as Vendor

-and-

IMAGE SENSING SYSTEMS, INC.

as Parent

### SHARE AND ASSET SALE AND PURCHASE AGREEMENT

in relation to

SHARES IN THE CAPITAL OF IMAGE SENSING SYSTEMS UK LIMITED

AND CERTAIN ASSETS OF THE IMAGE SENSING SYSTEMS GROUP

# COVINGTON

5

5

13

14

14

15

15

16

16

16

16

16

17

17

17

17

17

17

18

18

20

20 21

21

21

21

21

21

24

24

**25** 25

25

25

26

26

27

28

28

28

28

28

28

28 29

29

### TABLE OF CONTENTS

1. DEFINITIONS AND INT	TERPRETATION
------------------------	--------------

- 1.1 Definitions
  - 1.2 Interpretation
- SALE AND PURCHASE OF TARGET SHARES
- 2.1 Sale and purchase of Target Shares
- 2.2 No partial sale
- 2.3 Voting power pending registration
- 2.4 Stamp duty

#### 3. SALE AND PURCHASE OF LPR ASSETS

4. PRICE

2.

- 4.1 Purchase price
- 4.2 Attribution of Price
- 4.3 Payment of Price

#### 5. COMPLETION

- 5.1 Completion meeting
- 5.2 Completion obligations
- 6. Escrow
  - 6.1 General right of recourse
  - 6.2 Liability for claims satisfied from Escrow
  - 6.3 Operation of Escrow
  - 6.4 Release of cash
  - 6.5 Failure to release Escrow Amounts when due
  - 6.6 Adjustment of Price
  - 6.7 Agreement prevails
  - 6.8 Remittance

## 7. COMPLETION STATEMENT

- 7.1 Price adjustment
- 7.2 Preparation of Completion Statement
- 7.3 Payments due to Price adjustment
- 7.4 Accounting policies and practices
- 8. PARENT AND VENDOR ASSURANCES
  - 8.1 Transfer of Target Shares
  - 8.2 Assurances at Completion
  - 8.3 Restrictive covenants

## 9. PURCHASER WARRANTIES AND UNDERTAKINGS

- 9.1 Purchaser warranties
- 9.2 Access to Information

## **10. PARENT WARRANTIES**

- 10.1 Giving of Warranties
- 10.2 Disclosure
- 10.3 Limitations
- 10.4 Parent's knowledge and awareness
- 10.5 Warranties to be independent
- 10.6 Indemnity basis of recovery
- 10.7 Adjustment of Price
- 10.8 Calculation of liability

11.	LIMITA	TIONS ON LIABILITY	29
	11.1	Application of limitations	29
	11.2	Disapplication of limitations	29
12.	TAX IN	DEMNITY	29
	12.1	Fundamental Tax Warranties	29
13.	Existi	ng Lease	30
	13.1	Existing Lease	30
	13.2	Transitional Period	30
14.	CONFII	DENTIAL INFORMATION	30
	14.1	Non-disclosure of Confidential Information	30
	14.2	Exceptions	30
	14.3	Disclosure between members of the Receiving Group and/or Delegates	31
	14.4	Obligations survive termination	31
	14.5	Injunctive relief	31
	14.6	Publicity	31
15.	NOTICI		31
	15.1	Method of service	31
	15.2	Deemed service	32
	15.3	Proof of service	33
	15.4	Receipt outside business hours	34
	15.5	Change of address	34
	15.6	Service of proceedings	34
16.	MISCE	LLANEOUS	34
	16.1	Binding nature of this Agreement	34
	16.2	Further assurance	34
	16.3	Costs and other payments	34
	16.4	Assignment and other dealings	35
	16.5	Entire agreement	35
	16.6	Variation	35
	16.7	Severance	35
	16.8	Counterparts	36
	16.9	Language of the Agreement	36
	16.10	No agency, joint venture or partnership	36
			36
	16.12	Waiver	36
	16.13	Liability	37
		Rights and remedies are cumulative	37
		Obligations to remain in force notwithstanding Completion	37
		Third party rights	37
17.		NING LAW AND JURISDICTION	37
	17.1	English law	37
	17.2	Jurisdiction	37
	17.3	Process Agent	38
SCHE	DULE 1		39
		company information	39
SCHE	DULE 2		40
	Bank A	Accounts	40
	Part 1 -	Vendor's Lawyers' Client Account	40

SCHEDULE 3	41
Completion obligations	41
SCHEDULE 4	44
Warranties	44
Part A - Fundamental Warranties	44
Part B - Business Warranties	46
Part C - Tax Warranties	54
SCHEDULE 5	57
Limitations	57
SCHEDULE 6	62
Completion Statement	62
Part A - Pro-Forma Completion Statement	62
Part B - Estimated Completion Statement	64
Part C - Completion Statement accounting policies	66
SCHEDULE 7	68
2013 Audited Accounts	68
Schedule 8	76
Real Properties	76
Part A - Owned Properties	76
Part B - Leased Properties	76
SCHEDULE 9	77
LPR Assets	77
Part A - LPR Accounts Receivable	77
Part B - LPR Inventory	79
Part C - LPR IP	80
Part D - Customer List and Non-Compete	86
	00

5 of 78

THIS AGREEMENT is made by way of Deed on 9 July 2015.

#### **BETWEEN:**

- TAGMASTER AKTIEBOLAG, a public company incorporated in Sweden with company number 556487-4534 whose registered office is at Kronborgsgränd 11, 164 46 Kista, Sweden (the "Purchaser");
- 2. **IMAGE SENSING SYSTEMS EMEA LIMITED**, a private company incorporated in England and Wales with company number 08740702 whose registered office is at Unit 4 Caxton Place, Caxton Way, Stevenage, Hertfordshire, United Kingdom SG1 2UG (as the "Vendor"); and
- 3. **IMAGE SENSING SYSTEMS, INC.**, a corporation incorporated in Minnesota having its business address at 500 Spruce Tree Centre, 1600 University Ave. West, St. Paul, Minnesota 55104 USA (as the "**Parent**"), together the "**Parties**" and each a "**Party**".

#### WHEREAS:

- A. Image Sensing Systems UK Limited, a private company limited by shares incorporated in England and Wales with company number 03791347 (the "Target"), is a wholly owned subsidiary of the Vendor. Both the Vendor and the Target are subsidiaries of the Parent.
- **B.** The Vendor has agreed to sell (and the Parent has agreed to procure the sale of), and the Purchaser has agreed to purchase, the entire issued share capital of the Target and certain other assets in each case on the terms and subject to the conditions set out in this Agreement.
- C. In parallel with Vendor's sale of shares in the capital in the Target, the Parent has agreed to sell, and the Purchaser has agreed to purchase, certain assets owned by the Parent which primarily relate to the Vendor Group's license plate recognition business (the "**Business**") in each case on the terms set out in this Agreement.
- **D.** As part of a reorganisation carried out by the Vendor Group prior to the date of this Agreement, the Target has transferred to the Vendor certain assets and liabilities which do not primarily relate to the Business (the "**Reorganisation**").

## IT IS AGREED AS FOLLOWS:

1. **DEFINITIONS AND INTERPRETATION** 

#### 1.1 Definitions

1.1.1 In this Agreement, the following terms shall, unless the context otherwise requires, have the following meanings:

"Accounts" means the abbreviated accounts comprising the audited balance sheet of the Target made up to the Accounts Date, including the notes thereto and the auditors report thereon, and attached hereto as Schedule 7;

6 of 78

"Accounts Date" means 31 December 2013;

"Affiliate" means in respect of any undertaking, its subsidiary undertakings and parent undertakings and the subsidiary undertakings of each such parent undertaking, as well any other person at any time directly or indirectly controlling, controlled by, or under common control with, such undertaking; provided, however, that, for purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such person, whether through the ownership of voting securities, by contract or otherwise;

"Agreed Form" means, in relation to any document, the form of that document most recently approved by the Purchaser and the Parent and which has been initialled for the purpose of identification by the Purchaser's Lawyers and Vendor's Lawyers (or otherwise by or on behalf of the Purchaser and the Parent);

"Agreement" means this agreement, together with all its recitals and Schedules;

"Associate" means, in respect of a person, any Affiliate of that person and any other person connected with that person;

"Authorisations" means any licence, consent, permit, approval, certificate, clearance or other authorisation or confirmation of compliance, whether public or private granted by any Authority or other person;

"Authority" means any governmental, administrative, supervisory, regulatory or determinative authority, agency, court or other organisation of any jurisdiction, in each case which is established by, or having the authority of, Law;

"Business" is defined in Recital C;

"Business Day" means any day other than a Saturday or Sunday or a day which is a public holiday in London, England or Stockholm, Sweden;

"CA 2006" means the Companies Act 2006;

"CAA 2001" means the Capital Allowances Act 2001;

"CitySync Intangibles" means the customer list and non-compete set out in Part D of Schedule 9

"Completion" means completion of the sale and purchase of the Target Shares and the LPR Assets pursuant to this Agreement;

"Completion Date" is defined in clause 5.1;

"Completion Net Asset Value" means the net asset value shown in the Completion Statement;

"Completion Statement" means the final completion statement as prepared and determined in accordance with clause 7;

"Confidential Information" means information of a confidential nature (regardless of whether or not such information is recorded in any physical, electronic or other media and including all copies, extracts and derivations of confidential information) concerning the business, affairs, assets and/or interests of the Target and/or the Business, including technical data, know-how, designs, plans, specifications, methods, processes, controls, systems, trade secrets, recipes, formulae, research and development data, product complaint and testing information, lists of customers and suppliers, information relating to development, engineering, manufacturing, marketing, distribution, sale or purchase of goods and/or services, accounts, financial statements, financial forecasts, business plans, budgets, estimates, sales information, other financial information and any other information concerning the business, affairs, assets and/or interests of the Target and/or the Business which is marked as being confidential or would reasonably be expected to be kept confidential;

"Consequential Loss" means consequential, special, indirect, incidental or punitive damages or loss or any loss of profits or other form of economic loss, whether the claim in respect of such loss arises in contract, in tort or on any other basis;

"CRTPA 1999" means the Contracts (Rights of Third Party) Act 1999;

"CTA 2010" means the Corporation Tax Act 2010;

"Defaulting Party" is defined in clause 6.5.1;

"Delegates" means, in respect of any undertaking, the officers, employees, consultants, auditors, insurers, members, finance providers and professional advisers of such undertaking;

"Disclosure Bundle" means the bundle of documents (in an Agreed Form) provided by the Parent to the Purchaser in connection with the making of certain disclosures against the Warranties;

"Disclosure Letter" means the letter (in an Agreed Form) from the Parent to the Purchaser and dated the date of the Agreement and making certain disclosures against the Warranties, together with the Disclosure Bundle;

"**Dispute**" means any dispute arising out of, relating to or in connection with, this Agreement, including any question relating to the existence, validity, interpretation or termination of this Agreement or to any contractual or non-contractual obligation related to the Agreement and any dispute relating to the enforcement of the Agreement;

"Disputed Matters" has the meaning given to it in clause 7.2.5;

"Draft Completion Statement" has the meaning given to it in clause 7.2.1;

"Encumbrance" includes any interest or equity of any person (including any right to acquire, option or right of pre-emption); any mortgage, charge, pledge, lien, assignment, hypothecation, security interest (including any created by Law), title retention or other security interest arising under any agreement or arrangement for rental, hire purchase, credit or conditional sale or other agreement for payment on deferred terms;

"Enlarged Group" means the Purchaser and its Affiliates (including, with effect from Completion, the Target (for so long as the Target remains an Affiliate of the Purchaser)) and "member of the Enlarged Group" means any one of them;

"Escrow" means the escrow arrangements the subject of the Escrow Agreement and clause 6;

"Escrow Account" means the escrow account to be established and maintained in accordance with clause 6 and the Escrow Agreement;

"Escrow Agent" means the Purchaser's UK Lawyers and the Vendor's Lawyers, in their capacity as joint escrow agents under the terms of the Escrow Agreement;

"Escrow Agreement" means the agreement (in an Agreed Form) between the Parent, the Purchaser, the Purchaser's UK Lawyers and the Vendor's Lawyers and concerning, amongst other matters, the operation of the Escrow Account;

"Escrow Amount" means the remaining balance of the principal amount of cash held in Escrow (being initially that amount paid into the Vendor's Lawyers' Client Account (pending the opening of the Escrow Account) pursuant to clause 4.3(b)) (excluding accrued Income (if any) thereon);

"Escrow Period" means the period commencing on the Completion Date and ending on (and including) the date 12 months thereafter;

"Escrow Release Objection Notice" is defined in clause 6.5.1(d);

"Estimated Completion Statement" means the estimated completion statement set out in Part 2 of Schedule 6;

"Estimated Net Asset Value" means the net asset value shown in the Estimated Completion Statement;

"Existing Lease" has the meaning given in clause 13.1;

"Foreign Party" is defined in clause 17.3.1;

"Fundamental Tax Warranties" means the Tax Warranties set out at paragraphs 1.1.1 to 1.1.5 (inclusive) in Part C of Schedule 4;

"Fundamental Warranties" means those Warranties set out in Part A of Schedule 4;

"Group" means, in respect of any undertaking, such undertaking together with its Affiliates;

"Hardware" means the computer and data processing systems used by the Target in the Business excluding the Software;

"HMRC" means HM Revenue & Customs;

"ITEPA 2003" means the Income Tax (Earnings and Pensions) Act 2003;

"**Income**" is defined in clause 6.3.1(b);

"Independent Accountants" shall have the meaning given to it in clause 7.2.5;

"Intellectual Property Right" includes all patents, trademarks, service marks, domain names, rights (registered or unregistered) in any designs; applications for any of the foregoing; proprietary trade or business names, copyright (including rights in computer software), rights in databases, rights in inventions, proprietary know-how, rights in and rights to protect the use and/or disclosure of information, and all proprietary rights and forms of protection of a similar nature to any of the foregoing (including all applications, registrations, renewals and extensions in respect thereof) and all other forms of intellectual property rights, in each case in any jurisdiction;

"Internationally Recognised Stock Exchange" means any market in securities operated by an investment exchange which is a 'Recognised Investment Exchange', 'Recognised Overseas Investment Exchange' or 'Designated Investment Exchange' (in each case as so designated by the UK Financial Conduct Authority from time to time);

"**IPR Agreements**" means the written agreements listed in the Disclosure Letter pursuant to which the Target grants rights to third parties to use Intellectual Property Rights of which it is the owner or pursuant to which the Target is granted rights to use Intellectual Property Rights owned by a third party;

"IT Systems" means the Hardware and the Software;

"Law" includes all civil codes, statutes, regulations, rules of common law, judgments, decrees or orders of any Authority and other measures or decisions having the force of law in any jurisdiction from time to time, whether before or after the date of this Agreement;

"Leased Premises" means the real properties leased by the Target, as detailed in Part B of Schedule 8;

"Losses" means actions, proceedings, losses, damages, liabilities, claims, costs and expenses, including fines, penalties, legal and other professional fees and expenses and whether pursuant to a claim for contribution or under statute, contract, tort or otherwise;

"LPR" means license plate recognition;

"LPR Accounts Receivable" means the accounts receivable listed in Part A of Schedule 9;

"LPR Assets" means the LPR IP, the LPR Accounts Receivable and the LPR Inventory;

"LPR Inventory" means the inventory listed in Part B of Schedule 9;

"LPR IP" means the Intellectual Property Rights listed in Part C of Schedule 9;

"Management Accounts" means the unaudited management accounts of the Target as at 31 March 2015;.

"Material Contracts" means the Material Customer Contracts and the Material Supplier Contracts;

"Material Customer" means the customers identified as Material Customers in the Disclosure Letter;

"Material Customer Contracts" means the agreements entered into with the Material Customers.

"Material Supplier" means the suppliers identified as Material Suppliers in the Disclosure Letter;.

"Material Supplier Contracts" means the agreements entered into with the Material Suppliers

"Net Asset Shortfall" means the amount (if any) by which the Completion Net Asset Value is less than the Estimated Net Asset Value, as shown in the Completion Statement;

"Non-defaulting Party" is defined in clause 6.5.1;

"Objection Notice" has the meaning given to it in clause 7.2.1;

"Owned Premises" means the real properties owned by the Target, as detailed in Part A of Schedule 8;

"Permitted Purpose" means the bona fide implementation, pursuance and enforcement of this Agreement and other Transaction Documents and the undertaking of such other ancillary matters which are reasonably or necessarily undertaken in connection therewith;

"Premises" means the Owned Premises and the Leased Premises;

"Price" is defined in clause 4.1.1;

"Proceedings" is defined in clause 17.2;

"Process Agent" is defined in clause 17.3.1;

"Pro-Forma Completion Statement" means the form of completion statement set out in Part 1 of

Schedule 6;

"Purchaser Group Documents" is defined in clause 9.1.1(a);

"Purchaser's Lawyers" means Advokatfirman Lindahl of Mäster Samuelsgatan 20, 111 44 Stockholm, Sweden;

"Purchaser's UK Lawyers" means CMS Cameron McKenna LLP of Cannon Place, 78 Cannon Street, London EC4N 6AF;

"Receiving Group" is defined in clause 14.1;

"Receiving Party" is defined in clause 14.1;

"Relevant Securities" means, in respect of any undertaking:

(a) any share, security or other interest in the capital of such undertaking from time to time; and

(b) any other security, option, warrant, agreement or instrument which confers any right to subscribe, exchange for, convert into or otherwise acquire any issue of any share, security or other interest in the capital of such undertaking;

"Relief" means any loss, relief, allowance, exemption, set-off, right to repayment or credit (together with any related interest) in respect of any Tax or any deduction in computing income, profits or gains for the purposes of any Tax;

"**Restricted Business**" means the business as conducted by the Target as at Completion being the development and manufacture of LPR (also known as automatic number plate recognition) software and hardware, including the supply of LPR cameras, processor units, other hardware components and complete LPR systems, and any business which competes (or which is intended to, or reasonably likely to, compete) with such business of the Target and including (without limitation) the development and sale of automatic number plate recognition hardware;

"Restricted Period" means the period of 18 months as from the date of Completion;

"Software" means all computer software used by the Target in the Business at the date of this Agreement (excluding any "off-the-shelf" and "shrink wrap" software);

"Target" has the meaning given to it in Recital A;

"Target Shares" means the shares in the capital of the Target;

"**Tax**" means all forms of taxation, duties, levies, imposts, contributions, withholdings or amounts in the nature thereof whether of the United Kingdom or any other jurisdiction collected or assessed by, or payable to, any relevant Taxation Authority (irrespective of the person to which any such taxation, duties, levies, imposts, contributions, withholdings or amounts are directly or primarily chargeable)

and all penalties, fines, charges and interest incidental or relating to any of the foregoing or resulting from a failure to comply with the provisions of any Law relating to taxation (and "**Taxation**" shall be construed accordingly);

"Taxation Authority" means any government, state or municipality or any local, state, federal or other fiscal, revenue, customs or excise authority, body or official competent to impose, administer, levy, assess or collect Tax in the United Kingdom or elsewhere;

"Tax Claim" means a claim for breach of any of the Tax Warranties;

"Tax Warranties" means the Warranties set out in Part C of Schedule 4;

- "TCGA 1992" means the Taxation of Chargeable Gains Act 1992;
- "TIOPA 2010" means the Taxation (International and Other Provisions) Act 2010;
- "Transaction Documents" is defined in clause 16.5.1;

"US GAAP" means the generally accepted accounting principles of the United States of America as in effect on the Completion Date.

"VAT" means value added tax or any other sales tax that is levied on the supply of goods or services in any jurisdiction;

"VATA 1994" means the Value Added Tax Act 1994;

"Vendor Group" means the Parent together with its subsidiary undertakings as at the date of (but prior to) Completion (and "Vendor Group Company" shall mean any member of the Vendor Group);

"Vendor Group Documents" is defined in Part A of Schedule 4;

"Vendor's Lawyers" means Covington & Burling LLP of 265 Strand, London, United Kingdom WC2R 1BH;

"Vendor's Lawyers' Client Account" means the client account of the Vendor's Lawyers, details of which are set out in Part 1 of Schedule 6;

"Warranties" means the warranties set out in Schedule 4, at clause 10.1, and at clause 12.1;

"Warranty Claim" means a claim for breach of any of the Warranties as given at clause 10.1 and at clause 12.1;

## 1.2 Interpretation

- 1.2.1 Unless the context otherwise requires, the following rules of interpretation shall apply to this Agreement:
  - (a) words and phrases, the definitions of which are contained or referred to in the Companies Act 2006, shall have the meanings respectively attributed to them by it;
  - (b) words in the singular include the plural and in the plural include the singular;
  - (c) use of any gender or neuter includes the other genders and neuter;
  - (d) references to a particular statute or statutory provision or other Law shall:
    - (i) include all subordinate legislation made from time to time under that statute, statutory provision or other Law; and
    - (ii) be construed as a reference to such Law as amended, re-enacted, consolidated, supplemented, replaced or renumbered (or as its application or interpretation is changed or affected by other Laws) from time to time and as was, is, or will be (as the case may be) applicable at the time in question except that as between the Parties, no such amendment or modification shall apply for the purposes of this Agreement to the extent that it would impose any new or extended liability, obligation or restriction on, or otherwise adversely affect the rights of, any Party;
  - (e) references to "**this Agreement**" or any other agreement, deed or instrument is a reference to this Agreement or, as the case may be, the relevant agreement, deed or instrument as amended, supplemented, replaced or novated from time to time;
  - (f) references to "clauses" and "Schedules" are to clauses of, and schedules to, this Agreement;
  - (g) references to a "**paragraph**" or a "**Part**" are to a paragraph or part of the Schedule in which such reference appears;
  - (h) the expression "**connected**" with reference to a person or group of persons has the meaning given to it in sections 1122 and 1123 of the Corporation Tax Act 2010 (subject to the deletion of section 1122(8));
  - (i) references to a "day" shall mean a period of 24 hours running from midnight to midnight and reference to any time or date shall, save where otherwise expressly stated to the contrary, be a reference to the time or date (as the case may be) in London, England;
  - (j) references to a "**person**" shall be construed so as to include:
    - (i) any individual, firm, body corporate, Authority, joint venture, association, undertaking, partnership or limited partnership (whether or not having separate legal personality); and
    - (ii) a reference to the estate, successors, permitted transferees and permitted assignees of any of such person;
  - (k) the words "include", "including" or "in particular" shall not limit the generality of any preceding words or be construed as being limited to the same class as any preceding words where a wider construction is possible;

- (l) references to "written" or "writing" shall include all data in written form whether represented in hand-written, facsimile, printed, electronic or other format (including e-mail);
- (m) references to any English legal term for any action, remedy, method of judicial proceedings, legal document, legal status, court, official or any legal concept or thing shall in respect of any jurisdiction (other than England) shall be treated as a reference to any analogous term in that jurisdiction;
- (n) references to the "**ordinary course of business**" of the Target refers to the course of conduct of the business of the Target, having regard to its nature, size and scale, customary practices and policies and manner of operations, as current at the date of this Agreement; and
- 1.2.2 The Table of Contents, headings and titles are for convenience only and do not affect the interpretation of this Agreement.
- 1.2.3 The Schedules form part of this Agreement and shall have the same force and effect as if expressly set out in the body of this Agreement, and any reference to this Agreement shall include the Schedules.
- 1.2.4 If any provisions of the Schedules at any time conflict with any of the other provisions of this Agreement (not contained in the Schedules), the provisions of this Agreement (not contained in the Schedules) shall prevail.

#### 2. SALE AND PURCHASE OF TARGET SHARES

### 2.1 Sale and purchase of Target Shares

At Completion, the Vendor shall sell and the Parent shall procure that the Vendor shall sell, fully paid up and free from all Encumbrances and together with all rights now and in the future attaching to them, all of the entire legal and beneficial interest in the Target Shares (including the right to receive dividends or distributions declared, paid or made on or after Completion), which comprise the entire issued share capital of the Target, and the Purchaser shall purchase all such Target Shares.

#### 2.2 No partial sale

The Purchaser shall have no obligation to complete the sale and purchase of the Target Shares pursuant to clause 2.1 unless the sale and purchase of all the Target Shares is completed simultaneously.

### 2.3 Voting power pending registration

- 2.3.1 With effect from Completion:
  - (a) the Vendor irrevocably and unconditionally appoints the Purchaser as its lawful agent and attorney to exercise all rights which the Vendor may have in its capacity as registered holder of Target Shares including (but not limited to) any right to:
    - (i) receive notice of, attend and vote at, any meeting of the shareholders of the Target (or any class thereof) or at any adjourned meeting of the shareholders of the Target (or any class thereof);

- (ii) nominate any proxy to attend and vote at any meeting of the shareholders of the Target (or any class thereof);
- (iii) consent to any meeting of the shareholders of the Target (or any class thereof) being held on short notice;
- (iv) consent to any variation of any rights attaching to any class of Target Share;
- (v) approve any resolution of the shareholders of the Target (or any class thereof) proposed by way of written resolution;
- (vi) deal with, and give directions in respect of, any dividends, moneys, securities, benefits, notices or communications (in whatever form) in respect of the Target Shares,

in each case as the Purchaser in its absolute discretion sees fit;

- (b) the Vendor undertakes:
  - (i) not to exercise any rights attaching to the Target Shares other than on the express written instruction of the Purchaser;
  - (ii) to hold on trust for the Purchaser and to promptly notify the Purchaser of anything received by it after Completion in its capacity as holder of any Target Shares, and to act promptly in accordance with the Purchaser's instructions in relation to any such thing; and
  - (iii) to ratify whatever the Purchaser lawfully does or causes to be done under the authority or purported authority of the power of attorney conferred by clause 2.3.1(a); and

- (c) the Purchaser shall:
  - (i) indemnify and hold the Vendor harmless against any act undertaken or thing done by the Purchaser pursuant to the authority or purported authority of the power of attorney conferred by the Vendor pursuant to clause 2.3.1(a); and
  - use its best endeavours to procure that it (or its nominee) is registered in the register of members of the Target as the legal holder of the Target Shares within 3 months of Completion.
- 2.3.2 The powers of attorney conferred by clause 2.3.1(a) are given to secure the proprietary interests of the Purchaser in the Target Shares with effect from Completion and is irrevocable unless revoked with the prior written consent of the Purchaser. Such power of attorney shall terminate (without prejudice to anything done by the Purchaser before termination) on the date on which the Purchaser (or its nominee) is entered in the register of members of the Target as the legal holder of the Target Shares.
- 2.3.3 The Vendor may delegate one or more of the powers conferred on it by clause 2.3.1(a) to an officer or officers appointed for that purpose by its board of directors by resolution or otherwise.

### 2.4 Stamp duty

The Purchaser shall be responsible for the payment of all stamp duty (and, if applicable, stamp duty reserve tax) on this Agreement and the transfers in respect of the Target Shares to be delivered to the Purchaser at Completion.

#### 3. SALE AND PURCHASE OF LPR ASSETS

Subject to the terms of this Agreement, the Parent shall sell (with full title guarantee and free from all Encumbrances) and the Purchaser shall Purchase the LPR Assets, as at and with effect from Completion.

### 4. PRICE

### 4.1 Purchase price

- 4.1.1 Subject to adjustment in accordance with clause 7, the aggregate price (the "**Price**") payable by the Purchaser for the Target Shares, the LPR Assets, and the CitySync Intangibles shall be the sum of \$4,200,000, which shall be attributed between the Target Shares, the LPR Assets, and the CitySync Intangibles as set out in clause 4.2.
- 4.1.2 The Price shall be payable by the Purchaser to the Parent, on behalf of it and the Vendor, in cash on Completion in accordance with clause 4.3;

### 4.2 Attribution of Price

The Price shall be attributed between the Target Shares, the LPR Assets, and the CitySync Intangibles as follows:

(a) \$\* shall be payable by the Purchaser in respect of the Target Shares;

\*Certain portions denoted with an asterisk have been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.



- (b) \$\* shall be payable by the Purchaser in respect of the LPR Accounts Receivable;
- (c) \$\* shall be payable by the Purchaser in respect of the LPR Inventory and related reserves;
- (d) \$\* shall be payable by the Purchaser in respect of the LPR IP; and
- (e) \$\* shall be payable by the Purchaser in respect of the CitySync Intangibles,

provided that, in connection with the finalisation of the Completion Statement in accordance with clause 7, a different attribution of the Price may be agreed between the Parties in writing.

#### 4.3 Payment of Price

On Completion the Purchaser shall:

- (a) pay \$\* to the Vendor's Lawyers' Client Account for the benefit of the Parent;
- (b) pay \$\* to the Vendor's Lawyers' Client Account on behalf of the Parent (which amount shall be held by the Vendor's Lawyers in accordance with clause 6 pending the opening of the Escrow Account, and shall be transferred by the Vendor's Lawyers to the Escrow Account once opened); and
- (c) pay \$\* to the Vendor's Lawyers' Client Account on behalf of the Parent for retention in accordance with clause 7.3.2.

### 5. COMPLETION

### 5.1 Completion meeting

Completion shall occur at the offices of the Purchaser's Lawyers on the date of this Agreement (the "Completion Date").

## 5.2 Completion obligations

The provisions of Schedule 3 shall apply.

### 6. Escrow

### 6.1 General right of recourse

- 6.1.1 Without prejudice to any remedy available to the Purchaser, the Escrow Amount shall, if so required by the Purchaser, be made available to satisfy any claim(s) by the Purchaser against the Parent and, where applicable, the Vendor under, or in respect of any breach of, this Agreement.
- 6.1.2 Save in respect of a claim for a breach of any Fundamental Warranty, the liability of the Parent and, where applicable, the Vendor under this Agreement in respect of:

\*Certain portions denoted with an asterisk have been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- (a) any Warranty Claim; or
- (b) any other claim under this Agreement (other than claims under clause 7 for any payment as a result of a Net Asset Shortfall, which shall be satisfied and limited in accordance with clause 7.3)

shall be satisfied from, and limited to, the Escrow Amount and the sole remedy of the Purchaser under this Agreement in respect of each such claim shall be to the release of Escrow Amounts (if any) to the Purchaser from the Escrow.

### 6.2 Liability for claims satisfied from Escrow

- 6.2.1 The Vendor and the Parent shall have no liability to each other in respect of any liability satisfied from the Escrow Amount.
- 6.2.2 Clause 6.2.1, shall not apply so as to limit the liability of the Parent (or, where applicable, the Vendor) in respect of any fraud by it (or any of its Associates) or any remedy available to any other Party against it (or any of its Associates) in respect thereof.

#### 6.3 Operation of Escrow

- 6.3.1 The Escrow Amount shall be held in the Escrow on the terms of the Escrow Agreement and the following terms:
  - (a) any bank fees or other charges arising in respect of the Escrow shall be paid by the Purchaser and the Parent, who will each pay one half of such bank fees and charges;
  - (b) any interest or profit generated on the Escrow Amount (the "**Income**") shall be held in Escrow with the principal Escrow Amount on which such Income accrued and, on release of such principal Escrow Amount, such Income shall also be released to the Party or Parties entitled to such principal Escrow Amount;
  - (c) subject to clauses 6.3.1(d) and 6.3.3, at the end of the Escrow Period the Purchaser and the Parent shall issue joint written instructions to the Escrow Agent to release the remaining balance of the principal Escrow Amount (together with accrued Income thereon) to the Parent;
  - (d) to the extent that, prior to the expiry of the Escrow Period, the Purchaser has notified the Parent of any claim in respect of which it may be entitled to be satisfied from the Escrow Amount and (whether before or after the end of the Escrow Period) the amount of any such claim has:
    - (i) been agreed in writing between the Parent and the Purchaser (such agreement not to be unreasonably withheld or delayed following any oral agreement between them); or
    - (ii) been judicially determined by court of competent jurisdiction (from which no appeal is available (or has ceased to be available without any such appeal having been made)).

then the Purchaser and the Parent shall promptly upon such agreement or determination (and in any event within 14 days thereof) issue joint written instructions to the Escrow Agent to release to the Purchaser:

- (i) such part of the principal Escrow Amount as is equal to the amount of such claim as so agreed or determined (or, if less, the whole of the remaining principal Escrow Amount); and
- (ii) all Income which has accrued on such principal Escrow Amount;
- 6.3.2 to the extent that the liability for, or the quantum of, any claim notified by the Purchaser to the Parent prior to the expiry of the Escrow Period, and in respect of which claim the Purchaser may be entitled to be satisfied from the Escrow Amount, shall not have been agreed or determined (in a manner described in any of sub-clauses 6.3.1(d)(i) to 6.3.1(d)(ii) before the expiry of the Escrow Period, then:
  - (i) such principal Escrow Amount as is reasonably sufficient to satisfy all such claims in full (together with accrued Income on such principal Escrow Amount) (or such other amount as the Purchaser and the Parent may agree in writing) shall continue to be retained in the Escrow after expiry of the Escrow Period pending agreement or determination ((in a manner described in any of sub-clauses 6.3.1(d)(i) to 6.3.1(d)(ii)) of such claim; and
  - the balance of the Escrow Amount (and accrued Income thereon) (less all retained amounts determined in accordance with (i) above) shall be released in accordance with 6.3.1(c).

Promptly thereafter upon the agreement or determination (and in any event within 21 days thereof) of such a notified claim, the Parent and the Purchaser shall issue joint instructions to the Escrow Agent to release to the Purchaser:

- (i) such part the retained principal Escrow Amount as is equal to the amount of such claim as so agreed or determined (or, if less, the whole of the retained principal Escrow Amount); and
- (ii) all Income which has accrued on such retained principal Escrow Amount.

After expiry of the Escrow Period, once all such claims notified prior to the expiry of the Escrow Period have been so agreed or determined, the Parent and the Purchaser shall promptly issue joint instructions to the Escrow Agent to release the remaining balance of the principal Escrow Amount (together with accrued Income thereon) to the Parent.

- 6.3.3 For the purposes of clauses 6.3.1(d), if and to the extent that:
  - (a) the Purchaser gives unconditional and irrevocable written notice to the Parent that a claim (or part thereof) is withdrawn; or
  - (b) if legal proceedings commenced by the Purchaser in respect of a claim (or part thereof) are unconditionally and finally withdrawn or struckout (without possibility of re-instatement); or
  - (c) the Purchaser has no right to commence legal proceedings in respect of a claim (or part thereof),

then such claim (or, as the case may be, the relevant part thereof) shall be deemed to have been determined with no release of any Escrow Amount (or Income) due to the Purchaser in respect thereof.

#### 6.4 **Release of cash**

- 6.4.1 The principal value of the cash held in Escrow (ignoring any Income thereon) shall be used for the purpose of determining the Escrow Amount to be released in satisfaction of any claim.
- 6.4.2 All principal Escrow Amounts released from the Escrow to the Purchaser shall be paid in cash in USD.
- In the event that any principal Escrow Amount is due to be released to the Parent, any Income accrued on the principal Escrow Amount so released to 6.4.3 the Parent shall also be released to the Parent.

#### 6.5 Failure to release Escrow Amounts when due

- If the Parent or Purchaser breaches its obligation to so issue a joint written instruction to the Escrow Agent in accordance with clauses 6.3.1(c), 6.3.1 6.5.1 (d), or 6.5.2 (as the case may be) (such Party in default being a "Defaulting Party"), then the non-defaulting Party (being the Purchaser or Parent, as the case may be, the "Non-defaulting Party") may solely issue such instruction to the Escrow Agent provided that:
  - the Non-defaulting Party shall indemnify and keep fully indemnified the Defaulting Party from and against all Losses which may be (a) suffered or incurred and which arise directly or indirectly out of, or in connection with, any Escrow Amount and Income released by the Escrow Agent in compliance with such sole instruction of the Non-defaulting Party to the extent it is subsequently agreed or determined (in a manner described in any of sub-clauses 6.3.1(d)(i) to 6.3.1(d)(ii)) that such Non-defaulting Party was not entitled to the Escrow Amount and Income so released on such sole instruction of the Non-defaulting Party;
  - the Non-defaulting Party shall procure that a copy of such sole instruction is also then notified to the Defaulting Party; (b)
  - the sole instruction of the Non-defaulting Party shall instruct the Escrow Agent to postpone the release of the relevant Escrow Amounts for (c)a period of 14 days from the date of such sole instruction; and
  - (d) the Defaulting Party shall have a right to object to such sole instruction by written notice served during such 14 day period on the Escrow Agent and the Non-defaulting Party (an "Escrow Release Objection Notice") which notice shall state (on reasonable grounds) why the Defaulting Party does not agree that instruction is required to be given to the Escrow Agent in accordance with this Agreement. In the event that an Escrow Release Objection Notice is so served, the sole instruction of the Non-defaulting Party shall be deemed withdrawn pending agreement or further determination (in a manner described in any of sub-clauses

6.3.1(d)(i) to 6.3.1(d)(ii)) of the matter in dispute between the Defaulting Party and the Non-defaulting Party as stated in the Escrow Release Objection Notice.

### 6.6 Adjustment of Price

Any sum payable to the Purchaser from the Escrow Amount shall be treated as a reduction in the Price by an equal amount.

#### 6.7 Agreement prevails

In the event of any inconsistency between the provisions of this Agreement and the Escrow Agreement, the provisions of this Agreement shall prevail as between the Parties.

#### 6.8 Remittance

From and after the Closing, if the Seller or its Affiliates receives or collects any funds relating to any LPR Assets, the Seller or its Affiliate shall remit such funds to the Purchaser within five Business Days after its receipt thereof. From and after the Closing, if the Purchaser or its Affiliates receives or collects any funds relating to an asset transferred in the Reorganisation, the Purchaser or its Affiliate shall remit any such funds to the Seller within five Business Days after its receipt thereof.

### 7. COMPLETION STATEMENT

### 7.1 Price adjustment

7.1.1 Without prejudice to any other adjustments to the Price which may be required pursuant to this Agreement, the Price as stated in clause 4.1 shall, subject to the further provisions of this clause 7, be reduced by an amount equal to the Net Asset Shortfall (if any);

provided that no adjustment shall be so made to the Price where the net reduction as so determined would be equal to or less than \$\* (but if in excess of \$\* then the whole of such adjustment shall be made, not merely the excess). Payments in consequence of any adjustment in Price pursuant to this clause 7.1.1 shall be paid and limited in accordance with clause 7.3.

- 7.1.2 References to the Price or the final Price in this clause 7 shall be to the Price before any further adjustments which may be required pursuant to the further provisions of this Agreement (other than this clause 7).
- 7.1.3 For the avoidance of doubt, any adjustment of the Price pursuant to this clause 7 shall be deemed to adjust only the portion of the Price payable by the Purchaser in respect of the Target Shares pursuant to clause 4.2(a), and not any portion of the Price payable by the Purchaser in respect of any of the LPR Assets.

### 7.2 Preparation of Completion Statement

7.2.1 The Purchaser shall prepare and submit to the Parent, not later than 60 days after Completion, a draft financial statement (the "**Draft Completion Statement**") showing the Purchaser's calculation of the Price by reference to the financial position of the Target as at the close of business on the Completion Date and stating the proposed Net Asset Shortfall

\*Certain portions denoted with an asterisk have been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.



(if any). Such statement shall be prepared on the basis of, and in accordance with, the requirements, accounting policies and accounting practices specified in clause 7.4 and presented in substantially the same form, and shall include each of the same line-items, as set out in the Pro Forma Completion Statement.

- 7.2.2 Within 30 days starting on the day after the Parent receives the Purchaser's Draft Completion Statement, the Parent shall notify the Purchaser whether the Parent agrees with the Draft Completion Statement and, if the Parent does not agree with it, such notification (an "**Objection Notice**") shall include a schedule setting forth those amounts or items with which the Parent disagrees and, to the extent reasonably possible, the quantum of the disagreement together with the reasons why the Parent disagrees with such amounts and any initial proposed adjustments to the Draft Completion Statement.
- 7.2.3 In order to enable the Parent to review the Draft Completion Statement, the Purchaser shall procure that there is made available to the Parent (at the offices of the Target), the following documents and information in so far as the same is in the possession or control of the Enlarged Group and reasonably required in connection with such review of the Draft Completion Statement:
  - (a) books of account and records relating to the Target;
  - (b) formal working papers (if any) relating to the preparation of the Draft Completion Statement; and
  - (c) such further information as may be reasonably requested by the Parent.
- 7.2.4 If the Parent shall notify the Purchaser of its agreement to the Draft Completion Statement or shall take no action either to notify such agreement or to serve an Objection Notice within the period of 30 days referred to in clause 7.2.2 or, following the service of an Objection Notice, the Parent and the Purchaser shall agree in writing on the Draft Completion Statement (with or without amendment), then the same shall become binding upon the Parties in its original (or, if so agreed in writing by the Parent and the Purchaser, its amended) form and shall then constitute the final Completion Statement for the purposes of this Agreement and the Price shall be adjusted (if applicable) in accordance with clause 7.1.1.
- 7.2.5 If the Parent shall within the period of 30 days referred to in clause 7.2.2 serve an Objection Notice, the Parent and the Purchaser shall, each acting reasonably and in good faith, enter into discussions and endeavour to agree (and, if so agreed, amend) the Draft Completion Statement, which shall then be the final Completion Statement. If the Parent and the Purchaser shall fail to agree on the Draft Completion Statement within 30 days of the service of the Objection Notice on the Purchaser, then either the Parent or the Purchaser may thereafter refer the matters in dispute (the "Disputed Matters") to such independent firm of accountants as they agree, or should they fail to reach such agreement on an independent firm of accountants within 21 days, the Disputed Matters shall be referred to such independent firm of chartered accountants (being a reputable accountancy firm with at least 100 partners with experience of providing accountancy services to companies whose business and geographic scope are similar to that of the Target) nominated by such third party as the Purchaser and the Parent may agree (or, in the absence of agreement, nominated (acting reasonably and in good faith) by the then president of the Institute of Chartered Accountants in England and Wales on application of either the Purchaser or the Parent). The independent firm of accountants, as so determined, (the "Independent Accountants") shall decide the Disputed Matters as experts and not as arbitrators and its decision shall be final and binding upon the Parties (save in the case of fraud or manifest material error).

- 7.2.6 The Independent Accountants shall act on the following basis, and the Parent and the Purchaser shall ensure that the terms of engagement of the Independent Accountants shall conform with such basis and shall otherwise be on such terms as they shall agree or, in the absence of such agreement, as may be determined by the Independent Accountants:
  - (a) their sole duty will be to determine the dispute between the Parent and the Purchaser referred to them under clause 7.2.5, which may include any dispute concerning the interpretation of this Agreement which may affect the content of the Completion Statement or any dispute as to the jurisdiction of the Independent Accountants to decide the dispute or the content or interpretation of their terms of reference;
  - (b) the Parent and the Purchaser shall each provide the Independent Accountants with all information relating to the Target which is in their possession or under their control (including, in the case of the Purchaser, information in the possession or control of the Enlarged Group and/or the Purchaser's accountants) and which the Independent Accountants may reasonably require (provided that where information prepared by the Purchaser's accountants is to be disclosed information, it is agreed and acknowledged that such disclosure may, if so required by the Purchaser's accountants, be conditional on their signing a non-disclosure letter in a customary form in favour of the Purchaser's accountants);
  - (c) the fees and expenses of the Independent Accountants (including the fees and expenses of any professional advisers engaged under clause 7.2.6(e)) shall be paid by the Parent and the Purchaser in equal proportions;
  - (d) the Independent Accountants shall determine their own procedure but shall:
    - (i) give the Parent and the Purchaser a reasonable opportunity to make written and, if agreed by the Independent Accountants, oral representations to them;
    - (ii) require that the Parent and the Purchaser supply each other with copies of their written representations at the same time as they are submitted to the Independent Accountants and allow the other to respond in writing to any such written representation; and
    - (iii) where reasonably practicable, permit the Parent and the Purchaser to be present when the other of them is making oral submissions to the Independent Accountants and to make oral submissions in reply;
  - (e) the Independent Accountants may, if they reasonably consider this necessary to reach their determination, instruct valuers, solicitors and other professional advisers;
  - (f) subject to the Independent Accountants having promptly upon request received from the Parent and the Purchaser all information referred to in clause 7.2.6(b) and the procedure referred to in clause 7.2.6(d) having been concluded, the

Independent Accountants shall make their determination as soon as is reasonably practicable but no later than 30 days (or such period as the Purchaser and the Parent, each acting reasonably, may agree) after the dispute is referred to them; and

(g) the Independent Accountants shall make their determination in writing and shall attach a list of all adjustments (if any) to be made to the Draft Completion Statement, and the Draft Completion Statement as so adjusted (if so adjusted), which shall become the final Completion Statement.

### 7.3 Payments due to Price adjustment

- 7.3.1 If by reference to the final Completion Statement the Price is to be decreased in accordance with clause 7.1.1, then an amount equal to the Net Asset Shortfall shall be repaid to the Purchaser by the Parent. Such payment shall be satisfied from the Vendor's Lawyers' Client Account and the sole recourse of the Purchaser in respect of any price adjustment under this clause 7 shall be to the release of funds to the Purchaser from the Vendor's Lawyers' Client Account.
- 7.3.2 For the purpose of satisfying payments under clause 7.3.1, the Vendor's Lawyers shall retain in the Vendor's Lawyers' Client Account an amount of \$\* until the Completion Statement has been finalised in accordance with clause 7.2 and the payment under clause 7.3.1 has been made, if applicable. Any payment made in accordance with clause 7.3.1 shall be limited to a maximum amount of \$\*.
- 7.3.3 Immediately upon agreement or determination of the final Completion Statement and any price adjustment in accordance with clauses 7.1 and 7.2 above, the Parent and the Purchaser shall issue joint instructions to the Vendor's Lawyers instructing either:
  - (a) the release of the entire retained amount of \$\* to the Parent; or
  - (b) in the event that any price adjustment payment is to be made pursuant to clause 7.3.1 above:
    - (i) the release to the Purchaser of such portion of the retained \$\* as is equal to the Net Asset Shortfall; and
    - (ii) the release to the Parent of the such portion of the retained \$\* as remains following the release of funds to the Purchaser pursuant to clause 7.3.3(b)(i),

and the Vendor's Lawyers shall release the retained \$\* in accordance with such instructions.

### 7.4 Accounting policies and practices

The Completion Statement shall be prepared in accordance with the accounting principles and policies set out in Part C to Schedule 6. For the avoidance of doubt, the LPR Assets shall not be included within the scope of the Completion Statement.

<sup>\*</sup> Certain portions denoted with an asterisk have been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

#### 8. PARENT AND VENDOR ASSURANCES

#### 8.1 Transfer of Target Shares

The Vendor and the Parent each consent to, and waive all rights of pre-emption, rights of first refusal, veto rights and such other rights they may have as otherwise prohibit, restrict or impair, the transfer of the Target Shares to the Purchaser in accordance with this Agreement.

### 8.2 Assurances at Completion

- 8.2.1 At Completion, the Vendor and the Parent each waive any claims or rights or causes of action they may have (howsoever arising, whether known or unknown) against:
  - (a) the Target;
  - (b) each officer, employee, consultant, auditor and professional adviser of the Target, in so far as such claim, right or cause of action concerns the affairs of the Target or the Business;
  - (c) each other, in so far as such claim, right or cause of action concerns the affairs of the Target or the Business,

in each case, save where such claim, right or cause of action arises under, or in respect of any breach of, any Transaction Document;

8.2.2 Nothing in clause 8.2.1 shall be construed as a waiver of any right of any officer, employee or consultant of the Target in respect of any unpaid fees, salary, benefits and expenses under his/her terms of engagement since the immediately preceding monthly pay-roll date of the Target.

#### 8.3 Restrictive covenants

- 8.3.1 the Parent undertakes to the Purchaser that it shall not (and it shall procure that its Associates shall not) during the Restricted Period, either on his/her own account (or on the account of its Associate) or in conjunction with or for the benefit of any other person, directly or indirectly, in any capacity whatsoever (including whether acting as principal or as an officer, employee or consultant of a third party):
  - (a) carry on or be concerned, engaged or interested in any Restricted Business (other than as a holder of interests in securities admitted to trading on any Internationally Recognised Stock Exchange and representing (in aggregate together with the holdings of his/her Associates) less than \* per cent (\*%) in number of all securities of the same class and designation so admitted to trading);
  - (b) entice away (or endeavour to entice away) from their employment (or other engagement) by the Enlarged Group, or employ or engage (or offer to employ or engage) in any capacity (including as an officer, employee or consultant), any person who is an officer, employee or consultant of the Target at Completion.
- 8.3.2 The provisions of clause 8.3.1 (subject to the further provisions of this clause 8.3) shall, with the prior written consent of the Purchaser, be directly enforceable by each member of the Enlarged Group notwithstanding that they may not be a signatory to this Agreement.

<sup>\*</sup> Certain portions denoted with an asterisk have been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.



- 8.3.3 The Parties agree and acknowledge that each of the covenants contained in clause 8.3.1 is reasonable and necessary for the protection of the Purchaser's legitimate interests in the goodwill of the business of the Target. Each of the covenants contained in clause 8.3.1(a) and 8.3.1(b) shall be construed as a separate and independent covenant and shall be enforceable by the Purchaser (and, subject to clause 8.3.2, by each member of the Enlarged Group) separately and independently of any right to enforce any one or more of the other covenants contained in clause 8.3.1. If one or more of the covenants is held to be void or unenforceable, the validity of the remaining covenants shall not be affected. If any such covenant shall be found to be void or voidable but would be valid and enforceable if some part or parts of the restriction or covenant were deleted, such restriction or undertaking shall apply with such modification as may be necessary to make it valid and enforceable.
- 8.3.4 Without prejudice to any other remedy which may be available, the Purchaser (and, subject to clause 8.3.2, any member of the Enlarged Group) shall be entitled to seek injunctive or other equitable relief in relation to any breach or prospective breach of any covenant in clause 8.3.1.
- 8.3.5 No breach of clause 8.3.1(b) shall arise by reason of:
  - (a) any public advertisement (including, but not limited to, placing advertisements online (Internet) in journals and newspapers) being made inviting applications for any position of employment or other engagement;
  - (b) the receipt of responses to any such public advertisement or other enquiry as to the availability of any position of employment or other engagement;
  - (c) any person offering such employment or other engagement to any such respondent or enquirer; and
  - (d) any person thereafter employing or engaging any such respondent or enquirer,

provided always that:

- (i) such public advertisement was not intentionally and specifically directed at any officers, managers, employees or consultants of the Target; and
- (ii) no ancillary communication was made specifically by (or on behalf of, at the direction of, or with the approval or knowledge of) the Parent (or its Associate) to any officers, employees or consultants of the Target directing them to the fact that such public advertisement was so made or indicating that they should make any such enquiry.
- 8.3.6 The Purchaser (and the members of the Enlarged Group) shall have no right to bring any claim in respect of any breach of clause 8.3.1 unless and until Completion has occurred (where after a claim may be brought in reference to any breach thereof whether occurring before or after Completion).

### 9. PURCHASER WARRANTIES AND UNDERTAKINGS

## 9.1 Purchaser warranties

- 9.1.1 The Purchaser warrants to Parent that:
  - (a) it (and each of its Associates that is a party thereto) has the capacity and authority (without requiring the Authorisation of any Authority) to enter into and perform this Agreement and each other transfer, certificate, undertaking and agreement (if any) entered into (or to be

entered into) by it (or, as the case may be, its Associate) pursuant to this Agreement (together "Purchaser Group Documents");

- (b) each Purchaser Group Document will, if and when executed and delivered by it (or, as the case may be, its relevant Associate), constitute binding obligations enforceable against it (or, as the case may be, its Associate) in accordance with its terms; and
- (c) it is not insolvent or bankrupt under any Laws applicable to it, nor is it unable to pay its debts as they fall due, nor has any arrangement (whether by court proceedings or otherwise) been proposed under which its creditors (or any group of them) could receive less than the amounts due to them nor are any proceedings in relation to any compromise or arrangement with creditors, any winding up, bankruptcy or other insolvency proceedings concerning it (or any of its/his/her assets or interests) current, pending or threatened.
- 9.1.2 The warranties given by the Purchaser in clause 9.1.1 are given as at the date of this Agreement.
- 9.1.3 Without restricting the rights of the Parent or the ability of the Parent to claim damages on any basis available to it, the Purchaser undertakes to the Parent that if there is a breach of any of the warranties given by the Purchaser at clauses 9.1.1(a) to 9.1.1(c) (inclusive) or any other covenant or undertaking given by the Purchaser under this Agreement, the Purchaser will, on demand by the Parent, pay to the Parent or, (in the case of a liability to another person which has not been discharged) at the Parent's direction, the person to whom the liability has been incurred an amount equal to:
  - (a) any liability which the Parent (or any member of the Vendor Group) has incurred which it would not have incurred had such warranty, covenant or undertaking not been breached; and
  - (b) all costs and expenses (together with any VAT or comparable tax thereon) which may be incurred by the Parent (or any other member of the Vendor Group) as a result of or in connection with any breach of such warranty, covenant or undertaking.

### 9.2 Access to Information

After the Closing, the Purchaser shall, and shall cause its Affiliates to, grant or furnish the Parent and its representatives, counsel and accountants reasonable access, during normal business hours and upon prior written notice of at least three business days, to the personnel, properties, books and records of the Target in connection with (i) any Tax submission or return of the Parent (or any member of the Vendor Group) to the extent necessary to prepare such Tax submission or return or (ii) any matter for which indemnification is being sought pursuant to clause 10.6. The Purchaser shall deliver, at the Parents' expense, complete and

accurate copies of any books and records reasonably requested by the Parent in connection with this clause 9.2.

### 10. PARENT WARRANTIES

#### **10.1** Giving of Warranties

Subject always to clauses 10.2, 10.3 and 10.4, the Parent warrants to the Purchaser that each of the Warranties is true and accurate on the date of this Agreement. The Parties acknowledge and accept that the Purchaser is entering into this Agreement in reliance upon each of the Warranties.

### 10.2 Disclosure

10.2.1 The Warranties are given subject to matters fairly disclosed (with sufficient explanation and detail to enable the Purchaser to identify the nature and scope of the matter disclosed) in or under the terms of the Disclosure Letter.

#### 10.2.2 On the date of this Agreement:

- (a) the Parent shall deliver to the Purchaser the Disclosure Letter (together with a copy of the Disclosure Bundle in electronic format) duly executed by the Parent; and
- (b) the Purchaser shall deliver to the Parent an acknowledgement of receipt of the Disclosure Letter (and such copy of the Disclosure Bundle) duly signed by the Purchaser.

### 10.3 Limitations

Any liability in respect of a Warranty Claim shall be subject to the provisions of clauses 6 and 11.

### 10.4 Parent's knowledge and awareness

Where a Warranty is qualified by a reference (however expressed) to the knowledge or awareness of the Parent, the Parent shall be deemed to know or be aware of anything which is actually known to any of the following individuals:

- (a) \_\_\_\_\_;
- (b) ;
- (c) ; and
- (d)

#### 10.5 Warranties to be independent

Each of the Warranties shall be separate and independent and, save as otherwise expressly provided, shall not be limited by reference to any other Warranty.

### 10.6 Indemnity basis of recovery

Without restricting the rights of the Target or the Purchaser or the ability of the Purchaser to claim damages on any basis available to it, the Parent undertakes to the Purchaser that if there is a breach of any Warranty the Parent will, on demand by the Purchaser, pay to the Purchaser or, (in the case of a



liability to another person which has not been discharged) at the Purchaser's direction, the person to whom the liability has been incurred an amount equal to:

- (a) any liability which the Target or the Purchaser has incurred which it would not have incurred had the Warranty not been breached; and
- (b) all costs and expenses (together with any irrecoverable VAT or comparable tax thereon) which may be incurred by the Purchaser, the Target or the Enlarged Group as a result of or in connection with any breach of the Warranty.

### 10.7 Adjustment of Price

Any sum payable to the Purchaser in respect of any Warranty Claim shall be treated as a reduction in the Price by an equal amount.

### 10.8 Calculation of liability

For the sake of clarity it is specifically noted that any liability of the Parent or the Vendor hereunder shall, subject to the provisions of Schedule 5, be calculated on the basis of the actual Loss caused by the breach of a Warranty or other breach of the Agreement, as applicable, without regard to the attribution of the Price made under clause 4.2.

# 11. LIMITATIONS ON LIABILITY

### 11.1 Application of limitations

Subject to clause 11.2, the liability of the Parent and the Vendor in respect of any claim arising under, or for breach of, this Agreement shall be limited as set out in Schedule 5.

### 11.2 Disapplication of limitations

The provisions of clause 11.1 and Schedule 5 shall not apply so as to limit the liability of the Parent in respect of any fraud or willful misrepresentation by it (or any of its Associates) or any remedy available to the Purchaser against such person(s) in respect thereof.

### **12.** TAX INDEMNITY

### 12.1 Fundamental Tax Warranties

- 12.1.1 Without prejudice to the right of the Purchaser to claim on any other basis or take advantage of any other remedies available to it, if any of the Fundamental Tax Warranties is breached, the Vendor undertakes to pay to the Purchaser on demand subject to the provisions of Schedule 5:
  - (a) the Tax payable by the Target that the Target would not have been required to pay if the Fundamental Tax Warranty had not been breached; and
  - (b) all reasonable third-party costs and expenses (including, without limitation, legal and other professional fees and costs and expenses properly incurred by the Purchaser or the Target as a result of the breach of the Fundamental Tax Warranty.
- 12.1.2 A payment made in accordance with the above provisions shall include any amount necessary to ensure that, after Taxation of the payment, the Purchaser is left with the same amount it would have had if the payment was not subject to Taxation.

#### 13. EXISTING LEASE

### 13.1 Existing Lease

The Parent and the Purchaser each hereby agree and undertake to use reasonable endeavours to procure that, as soon as reasonably practicable following Completion, the lease dated 5 November 2014 among (1) \_\_\_\_\_\_; (2) the Target; and (3) the Parent in relation to \_\_\_\_\_\_(the "Existing Lease") be amended to the effect that the Parent will no longer be a party to the Existing Lease, and will have no rights or obligations thereunder. Such reasonable endeavours may include (but are not limited to) the Purchaser complying with any reasonable request made by \_\_\_\_\_\_\_ as a condition to the provision of their consent as landlord under the Existing Lease for the Purchaser to become a party to the Existing Lease and having the same obligations as the Parent currently has under the Lease.

#### **13.2** Transitional Period

- 13.2.1 Until such time as the Existing Lease has been amended pursuant to clause 13.1 above, the Parent agrees and undertakes that it shall not seek to exercise or enforce any of its rights as a tenant under the Existing Lease.
- 13.2.2 The Purchaser agrees and undertakes to indemnify the Parent against any claims for rent, service charge or any other sums due under the Existing Lease with effect from Completion which may be made against the Parent following Completion.

#### 14. CONFIDENTIAL INFORMATION

### 14.1 Non-disclosure of Confidential Information

Subject to Completion and to clauses 9.2 and 14.2, the Parent agrees and undertakes to (and to procure that his/her/its Associates (such person and his/her/its Associates being a "Receiving Party", and collectively, the "Receiving Group") shall) keep confidential and not disclose to any third party, nor use other than for a Permitted Purpose, any Confidential Information.

#### 14.2 Exceptions

- 14.2.1 Clause 14.2 shall not apply if and to the extent that:
  - (a) such Confidential Information is in the public domain (other than by reason of a breach of any obligation of confidentiality applicable to the relevant Receiving Group);
  - (b) such Confidential Information is disclosed to the relevant Receiving Group on a non-confidential basis by person(s) (other than any member of the Enlarged Group (or any person acting on behalf thereof)) in circumstances where the relevant Receiving Group reasonably believed that such disclosure was lawfully made without breach of any obligation of confidentiality by such person(s);
  - (c) a member of the Enlarged Group has consented in writing to such disclosure and/or use of such Confidential Information or has otherwise confirmed in writing that such Confidential Information is not confidential; or

- (d) such disclosure or use is required by Law, the rules of any investment exchange to which the relevant Receiving Party's Group is subject, or by any competent Authority having jurisdiction over the relevant Receiving Group.
- 14.2.2 If disclosure or use is to be made pursuant to clause 14.2.1(d) then, if permitted by Law, the relevant Receiving Party shall consult with the Purchaser reasonably in advance of such disclosure or use so as to permit the Enlarged Group reasonable opportunity to review and comment on such disclosure or intended use and, if so desired by the Purchaser, for the Enlarged Group to take any reasonable action to prevent or restrict such disclosure or use.

#### 14.3 Disclosure between members of the Receiving Group and/or Delegates

Notwithstanding clause 14.1, the disclosure of Confidential Information between members of the Receiving Group and/or Delegates of the Receiving Group shall be permitted, provided that:

- (a) such disclosure is restricted to those persons who reasonably need to know such information in connection with the Permitted Purpose or by the nature of their role as a Delegate of the relevant Receiving Group; and
- (b) the relevant Receiving Party shall procure that persons to whom Confidential Information is disclosed under this clause 14.3, and all other Delegates of its/his/her Receiving Group, shall comply with the obligations of confidentiality and restrictions on use applicable under this clause 14 in the same manner as such restrictions and obligations apply to such Receiving Party.

### 14.4 Obligations survive termination

The obligations of each Receiving Party contained in this clause 12 shall survive Completion and shall continue for a period of 5 years after Completion.

### 14.5 Injunctive relief

Each Party acknowledges that monetary damages alone may not be a sufficient remedy for any actual or threatened breach of this clause 12, that injunctive and specific performance or any other equitable relief may be available to the non-defaulting Party in respect of any such breach and that no proof of special damages shall be necessary for the enforcement of this clause 12. Such remedies shall be in addition to, and not in lieu or limitation of, any other remedy available to the non-defaulting Party under this Agreement or otherwise at Law or in equity.

### 14.6 Publicity

The Parties agree that a joint press release concerning the transactions contemplated by this Agreement shall be prepared and issued on a date to be agreed. Each Party undertakes not to disclose, without the approval of the other Party, any information regarding the transactions contemplated by this Agreement in a manner or at time not agreed between the Parties. Notwithstanding the foregoing, each Party shall be entitled to disclose such information as is required by law or stock exchange recommendations or other similar regulations to which the Party is subject, in which case the Party undertakes to reasonably confer with the other Party prior to such disclosure.

### 15. NOTICES

### 15.1 Method of service

	manus was se	ce or other communication given under this Agreement by any Party to the other Party shall be in writing (which shall include e-mail), signed in cript by or on behalf of the Party giving it (which includes a faxed or scanned manuscript signature or, in the case of e-mail, that the message nt from an e-mail address of the Party giving it (and which sender's e-mail address is one to which notices and other communications may also dly delivered to that Party under this clause 15.1)), in the English language and may be either:				
	(a)	delivered personally by hand; or				
	(b)	if sent from within the same jurisdiction in which the recipient's address is located, then sent by first class pre-paid post or courier (or, if sent from outside the jurisdiction in which the recipient's address is located, then sent by international courier); or				
	(c)	sent by facsimile; or				
	(d)	sent by email,				
	in each	a case addressed as follows:				
	<u>For th</u>	For the Purchaser:				
	Address:					
	E-mail	E-mail address:				
	For the	For the attention of:				
	For th	For the Vendor:				
	Addres	SS:				
	E-mail	address:				
	For the	e attention of:				
	<u>For th</u>	<u>e Parent:</u>				
	Addres	SS:				
	E-mail	address:				
	For the	e attention of:				
15.2	Deemo	ed service				
	Withou any ev	at prejudice to any earlier time at which a notice or other communication may be actually given and received, a properly addressed notice will in ent:				

08/11/2015 03:01 PM

32 of 78

152795 - FINAL

iss152795\_ex2-1.htm

(a) if personally delivered, be deemed to have been given and received upon delivery at the relevant address;

2	2
3	2

- (b) if posted to an address in the same jurisdiction as that from which it was sent by first class pre-paid post or courier (which courier advises of delivery within two Business Days), be deemed to have been given and received two Business Days after the date of posting;
- (c) if sent to an address in a different jurisdiction as that from which it was sent by international courier (which courier advises of delivery within seven Business Days), be deemed to have been given and received seven Business Days after the date of posting;
- (d) if sent by facsimile and a confirmatory successful transmission report is given by the transmitting device, be deemed to have been given and received on the date of transmission (or, if such day is not a Business Day, then the next Business Day); and
- (e) if sent by e-mail and no delivery failure is reported to or by the sender's e-mail server, be deemed to have been given and received on the date such e-mail was sent (or, if such day is not a Business Day, then the next Business Day).

### 15.3 **Proof of service**

In proving service, it shall be sufficient to prove that:

- (a) the envelope containing the notice or other communication was addressed to the address of the relevant Party as set out in clause 15.1 (or as otherwise notified by that Party pursuant to clause 15.5) and delivered either to that address or into the custody of the postal authorities as first class pre-paid post or custody of the courier, or international courier firm; or
- (b) the notice or other communication was transmitted in full by facsimile to the facsimile number of the relevant Party set out in clause 15.1 (or as otherwise notified by that Party pursuant to clause 15.5) (as evidenced by a confirmatory transmission report); or

(c) that the e-mail was correctly addressed and that no delivery failure was reported to or by the sender's e-mail server.

### 15.4 Receipt outside business hours

If receipt or deemed receipt of a notice or other communication occurs before 9.30 a.m. in the country of receipt on a Business Day, the notice or other communication shall be deemed to have been received at 9.30 a.m. (in the country of receipt) on that day. If deemed receipt occurs after 5.30 p.m. (in the country of receipt) on a Business Day or on a day which is not a Business Day, the notice or other communication shall be deemed to have been received at 9.30 a.m. (in the country of receipt) on the country of receipt) on the next Business Day.

### 15.5 Change of address

Any Party to this Agreement may give at least five Business Days' notice to the other Party to change its address or other details specified in clause 15.1.

### 15.6 Service of proceedings

This clause does not apply to the service of any documents relating to any Proceedings or where applicable, any arbitration or other method of dispute resolution.

### 16. MISCELLANEOUS

### 16.1 Binding nature of this Agreement

Each Party warrants and undertakes to each other Party that this Agreement, and each other Transaction Document to which it is a party, has been duly executed by it and comprises a valid and legally binding obligation enforceable against it in accordance with the terms of this Agreement.

### **16.2** Further assurance

- 16.2.1 Each Party shall (at its/his/her own expense) execute and deliver all such documents and do all such things as may be reasonably required from time to time to give full effect to the provisions of this Agreement in so far as the same are to be performed by such Party.
- 16.2.2 Without prejudice to any other rights or remedies of the Purchaser under this Agreement, if any Intellectual Property belonging to the Parent or any other Vendor Group Company and which relates exclusively to the Business has not been transferred to the Target or the Purchaser by virtue of the transactions entered into under or pursuant to this Agreement:
  - (a) the Parent shall, as soon as reasonably practicable at its own cost, transfer or procure the transfer of such Intellectual Property to the Target, or otherwise to the order of the Purchaser, for no consideration; and
  - (b) each Party shall provide such assistance to the other Party as it reasonably requires for the purposes of this clause 16.2.2.

#### 16.3 Costs and other payments

16.3.1 Save as expressly provided in this Agreement to the contrary, each Party shall be responsible for its own costs incurred in connection with the negotiation, preparation, execution and implementation by



it of this Agreement and of all Transaction Documents, provided that this clause 16.3 shall not prejudice the right of any Party to seek to recover its costs in any litigation or dispute resolution procedure which may arise out of this Agreement.

### 16.4 Assignment and other dealings

- 16.4.1 No Party may assign or otherwise transfer any of its rights or obligations under this Agreement.
- 16.4.2 This Agreement shall be binding on and inure to the benefit of the Parties and their respective successors and their permitted assignees. Subject to, and upon any succession or assignment permitted by this Agreement, any successor or permitted assignee shall, in its own right, be able to enforce any term of this Agreement in accordance with its terms as if it were a Party but until such time, any such successor or permitted assignee shall have no rights, whether as a third party, or otherwise.

### 16.5 Entire agreement

- 16.5.1 This Agreement (together with the other documents to be entered into in connection with this Agreement in the Agreed Form) (the "**Transaction Documents**") constitute the whole agreement and understanding between the Parties relating to the subject matter of the Transaction Documents and supersede and extinguish any previous agreement or arrangement between the Parties relating to the subject matter thereof and excludes any representation, warranty, promise, assurance or other undertaking implied by Law, custom or course of dealing.
- 16.5.2 Each Party acknowledges and agrees that:
  - (a) it has not relied on, or been induced to enter into, this Agreement by any representation (including any misrepresentation), warranty, statement, assurance, promise or undertaking of any kind except as expressly included in the Transaction Documents; and
  - (b) no Party shall be entitled to claim damages and/or terminate or rescind this Agreement by reason of any misrepresentation (other than a fraudulent misrepresentation) having been made to it by any person (whether a Party or not) at any time and upon which it has relied before entering into this Agreement.
- 16.5.3 Nothing in this clause 16.5 shall limit or exclude any liability or remedy for fraud or willful misconduct.

#### 16.6 Variation

This Agreement may be varied by agreement in writing signed by the Purchaser and the Parent.

16.7 Severance

- 16.7.1 If any provision of this Agreement (or part of a provision) is held by any court of competent jurisdiction to be invalid, unenforceable or illegal, such provision (or part) shall to that extent be deemed not to form part of this Agreement and the other provisions of this Agreement shall remain in force.
- 16.7.2 If any invalid, unenforceable or illegal provision would be valid, enforceable or legal if some part of it were deleted, the provision shall apply with whatever modification is necessary to give effect to the intention of the Parties.

### 16.8 Counterparts

- 16.8.1 This Agreement may be executed in any number of counterparts, each of which is deemed to be an original and which together have the same effect as if each Party had signed the same document.
- 16.8.2 This Agreement may be executed through the use of facsimile transmission, and a counterpart of this Agreement that contains the facsimile signature of a Party, which counterpart has been transmitted by facsimile transmission to the other Party at such facsimile number as such other Party shall request, shall constitute an executed counterpart of this Agreement.

#### 16.9 Language of the Agreement

The language of this Agreement is English and all documents, notices, waivers and all other written communications or otherwise between the Parties in connection with this Agreement shall be in English.

#### 16.10 No agency, joint venture or partnership

Nothing contained in this Agreement shall constitute or be deemed to constitute an association, joint venture or partnership between the Parties and no Party shall be, or be construed to be, under the terms of this Agreement the agent of the any other Party for any purpose or to have any authority to bind or incur any liability on behalf of any other Party, save as expressly provided in this Agreement.

### 16.11 Discretion

Except as expressly provided to the contrary in this Agreement (including any express obligation to act in good faith, fairly and/or reasonably (including any provision which does not allow a Party to unreasonably withhold, condition or delays its approval or consent)), a Party may exercise any discretion given to it under this Agreement in its absolute discretion and any other Party may not challenge the exercise of that discretion on the grounds that the Party did not exercise its discretion fairly or reasonably. Such exercise of discretion shall not be subject to the dispute resolution procedures in this Agreement.

### 16.12 Waiver

16.12.1 No failure to exercise, nor any delay in exercising, any right, power, privilege or remedy under this Agreement shall in any way impair or affect the exercise of such right, power or

privilege or remedy, or operate as a waiver of such right, power or privilege or remedy in whole or in part.

- 16.12.2 The waiver by any Party of any of its rights or remedies arising under this Agreement or by Law shall not constitute a continuation of that or any other right or remedy.
- 16.12.3 No single or partial exercise of any right, power, privilege or remedy under this Agreement shall preclude or restrict the further exercise of that or any other right, power, privilege or remedy.

### 16.13 Liability

Save as expressly provided to the contrary in this Agreement, all obligations contained in this Agreement and made by the Parties are given or made by each of them severally in relation to himself only and each of them shall only be liable for any loss or damage arising from its/his/her own breach of such obligations.

#### 16.14 Rights and remedies are cumulative

Save as expressly provided in this Agreement where any rights and remedies are limited, the rights, powers, privileges and remedies provided in this Agreement are cumulative and are not exclusive of any rights, powers, privileges or remedies provided by Law or otherwise.

#### 16.15 Obligations to remain in force notwithstanding Completion

Except to the extent that they have been fully performed and discharged and except where this Agreement expressly provides otherwise, all provisions of this Agreement (including all representations, warranties and indemnities) remain in full force and effect notwithstanding Completion and shall not be extinguished or affected by Completion.

### 16.16 Third party rights

- 16.16.1 Except as provided to the contrary in this Agreement (including under clauses 16.4.2 and 16.16.2), a person who is not a party to this Agreement shall not have any rights under the CRTPA 1999 to enforce any term of this Agreement.
- 16.16.2 The provisions of clauses 8.3 and 12 are made for the benefit of the Purchaser and the Enlarged Group and, accordingly, each Enlarged Group company may, with the written consent of the Purchaser, in its own right enforce those provisions (subject to the further terms of this Agreement) in accordance with the provisions of the CRTPA 1999, provided that this Agreement may be amended, varied, suspended, terminated or rescinded without the consent of any members of the Enlarged Group which is not otherwise a Party.

### 17. GOVERNING LAW AND JURISDICTION

### 17.1 English law

This Agreement and all matters arising from it (including any Dispute) shall be governed by, and construed in accordance with, the Laws of England and Wales.

### 17.2 Jurisdiction

Each Party irrevocably submits to the exclusive jurisdiction of the English courts in relation to any Dispute and waives any objection to proceedings in relation to any Dispute ("**Proceedings**") in such courts on the grounds of venue or on the grounds that the Proceedings have been brought in an inappropriate forum, provided that a judgment or order of the English courts may be enforced in any court of competent jurisdiction.

### 17.3 Process Agent

- 17.3.1 Any Party to this Agreement for the time being incorporated or resident outside England (a "Foreign Party") shall appoint and maintain an agent (a "Process Agent") with an address in England for service of process of any Proceedings in England and shall deliver to the other Party details of the Process Agent's address (in England) and a copy of the Process Agent's written acceptance of such appointment. Service upon the Process Agent shall be good service upon the Foreign Party appointing the Process Agent whether or not the process is forwarded to, or received by, the Foreign Party.
- 17.3.2 The Foreign Party shall notify the other Party in writing of any change in the address of the Process Agent within 15 Business Days of such change.
- 17.3.3 If for any reason the Process Agent ceases to act, resigns or no longer has an address in England, the Foreign Party who appointed him shall immediately appoint a substitute process agent with an address in England and shall deliver to the other Party within 10 Business Days details of such substitute agent's address (in England) and a copy of the substitute agent's written acceptance of such appointment.
- 17.3.4 If any Foreign Party fails to appoint and maintain a process agent or fails to notify the other Party of a change in the Process Agent pursuant to clause 17.3.2 or any substitute agent pursuant to clause 17.3.3, it shall be effective service for the other Party to serve process upon the last known address in England of the last known process agent of that Foreign Party notwithstanding that such process agent is no longer found at such address or has ceased to act.
- 17.3.5 Nothing in this Agreement shall affect the right of service of process in any other manner permitted by Law.
- 17.3.6 In accordance with clause 17.3.1:
  - (a) the Parent gives notice that it has appointed the process agent named below:

Name	Address

(b) the Purchaser gives notice that it has appointed the process agent named below:

Name	Address

# Target company information

# **Bank Accounts**

# Part 1 - Vendor's Lawyers' Client Account

#### **Completion obligations**

#### 1. OBLIGATIONS OF THE VENDOR AND THE PARENT

#### 1.1 Parent obligations

The Parent shall:

- (a) procure that on or prior to Completion:
  - (i) a board meeting of the Target has been held giving effect to the following matters:
    - (1) approving the transactions contemplated by this Agreement
    - (2) approval of the transfer at Completion of the Target Shares and (subject only to due stamping) to register in the register of members of the Target, the Purchaser as the holder of the Target Shares in the statutory books of the Target;
    - (3) approving and authorising the execution of such documents as are to be entered into by the Target in connection with the transactions contemplated by this Agreement;
    - (4) the approval of the resignations of \_\_\_\_\_\_ as directors of the Target with effect from Completion;
    - (5) approving the appointment of as a director of the Target with effect from Completion;
    - (6) approving the change of the Target's name to "CitySync Limited"; and
    - (7) resolving that, with effect from Completion, all existing bank mandates and signing authorities in respect of the banking facilities of the Target shall be revoked and new authorities shall be approved and on such terms as the Purchaser may direct.
- (b) procure that at Completion there be delivered to the Purchaser:
  - (i) two counterparts of this Agreement duly executed by each of the Vendor and the Parent;
  - (ii) two copies of the Disclosure Letter in an Agreed Form, duly executed by the Parent;
  - (iii) copies of each power of attorney (each in an Agreed Form) under which this Agreement (and/or any document entered into in connection with the transactions contemplated by this Agreement) has been entered into by any of the Vendor, Parent or other Vendor Group Company (if any);

- (iv) the notices of resignation of \_\_\_\_\_\_ as directors of the Target with effect from Completion (in an Agreed Form), duly executed by the relevant directors;
- (v) two counterparts of the Escrow Agreement (in an Agreed Form) duly executed by the Parent;
- (vi) a printed copy of the minutes of the meeting of the board of directors of the Target (in an Agreed Form) referred to in paragraph 1.1(a)(i) above, certified as a true copy by a director of the Target;
- (vii) all the statutory and other books (duly written up to date) of the Target (including all certificates of incorporation, articles of association and other constitutional documents, register of members, and common seal);
- (viii) original signed versions of the following documents relating to the Reorganisation (in each case in an Agreed Form), duly executed by each of the parties thereto:
  - (1) deed of assignment between the Target and the Vendor relating to video and radar assets and liabilities;
- (ix) two counterparts of a deed of assignment (in an Agreed Form) assigning the LPR Assets from the Parent to the Purchaser, each duly executed on behalf of the Parent.

### 1.2 Obligations of the Vendor

- 1.2.1 The Vendor shall procure that there be delivered to the Purchaser at or before Completion:
  - (i) in respect of the Target Shares:
    - (1) a stock transfer form (in an Agreed Form) duly executed by the Vendor in favour of the Purchaser;
    - (2) the original share certificate(s) issued by the Target (or, in respect of any missing share certificate, an indemnity in an Agreed Form duly executed by the registered holder);
    - (3) a written resolution of the Target approving the change of the Target's name to "CitySync Limited", duly executed by the Vendor as the sole shareholder of the Target; and
  - such other documents, waivers or consents as the Purchaser may reasonably require to enable the Purchaser to be registered as holders of the Target Shares.

## 1.3 Obligations of the Purchaser

The Purchaser shall procure that there be delivered to the Parent at or prior to Completion:

(a) two counterparts of this Agreement, duly executed on behalf of the Purchaser;

(b) two copies of the Disclosure Letter (in an Agreed Form) duly executed by the Purchaser in acknowledgement of receipt;

(c) a copy of each power of attorney (each in an Agreed Form) under which this Agreement (and/or any document entered into in connection with the transactions contemplated by this Agreement) has been entered into by the Purchaser or any Affiliate of the Purchaser (if any); and

(d) two counterparts of the Escrow Agreement (in an Agreed Form) duly executed by the Escrow Agent and the Purchaser.

#### Warranties

#### Part A - Fundamental Warranties

### 1. INCORPORATION AND COMPLIANCE

- 1.1 The particulars of the Target set out in Schedule 1 are true, complete and accurate and not misleading.
- 1.2 The Target is duly incorporated and validly existing as a private limited company under CA 2006.
- 1.3 The Target has no subsidiary undertakings and owns no equity interests in any other person, nor has it agreed to acquire the same.
- 1.4 Due compliance has been made with all the provisions of CA 2006, and all other legal requirements, in connection with:
  - (a) any allotment, issue, purchase or redemption of shares, debentures or other securities in the Target;
  - (b) any reduction of the share capital of the Target;
  - (c) any amendment to the memorandum or articles of association of the Target;
  - (d) the passing of any resolutions by the Target; and
  - (e) the payment of any dividends by the Target.

### 2. SHARE CAPITAL AND TITLE

- 2.1 The Target Shares have been fully paid for and constitute all the issued and outstanding shares in the capital of the Target and there are no options or rights to acquire any unissued shares in the capital of the Target or any agreement or commitment to give or create any.
- 2.2 The Vendor is the registered and sole owner of the Target Shares, and the transfer of such Target Shares to the Purchaser under this Agreement at Completion constitutes the transfer of the whole of the right, title and interest (including all legal title to, and all beneficial interest in) such Target Shares free from all Encumbrances.
- 2.3 There is not now existing nor is there any agreement to create any Encumbrance on or affecting the Target Shares or on unissued shares or other Relevant Securities of the Target.
- 2.4 Save for the Target Shares to be transferred to the Purchaser by the Vendor at Completion in accordance with this Agreement, none of the Parent, the Vendor or any member of the Vendor Group holds any right, title or interest in respect of any further Relevant Securities of the Target.
- 2.5 The Vendor has the right to exercise all voting and other rights over the Target Shares.

# **3. AUTHORITY**

- 3.1 The Vendor, the Parent (and each other member of the Vendor Group that is a party thereto) has the capacity and authority (without requiring the Authorisation of any Authority) to enter into and perform this Agreement and each other transfer, certificate, undertaking and agreement (if any) entered into (or to be entered into) by it (or, as the case may be, its relevant Associate) pursuant to this Agreement (together "Vendor Group Documents").
- 3.2 Each Vendor Group Document will, if and when executed and delivered by the parties thereto, constitute binding obligations enforceable in accordance with its terms against the Vendor, the Parent and/or each other member of the Vendor Group that is a party thereto.

# 4. BROKERS

4.1 No agent, broker, investment banker, person or firm acting in a similar capacity on behalf of or under the authority of the Vendor, the Parent or the Target is or will be entitled to any broker's or finder's fee or any other commission or similar fee, directly or indirectly, which is payable by the Target on account of any of the transactions contemplated under any of the Transaction Documents.

#### Part B - Business Warranties

### 1. CONSTITUTION AND REGISTERS

- 1.1 The copy of the articles of association and memorandum of association of the Target annexed to the Disclosure Letter is accurate and complete.
- 1.2 The register of members of the Target contains an accurate and complete record of the current members of the Target and the Target has not received any written notice that the register is incorrect or incomplete or should be rectified.
- 1.3 The statutory books (excluding the minute books) of the Target are up to date, in its possession or control and the Target has not received any written notice that they are incorrect or incomplete or should be rectified in any material respect.

## 2. SOLVENCY

- 2.1 No order has been made and no resolution has been passed for the winding up of the Target or for a provisional liquidator or manager to be appointed in respect of the Target and no petition has been presented and no meeting has been convened for the purposes of the winding up of the Target.
- 2.2 No administrator has been appointed in relation to the Target, no documents have been filed with the court for the appointment of an administrator of the Target and no notice of an intention to appoint an administrator has been given by the Target or any of its directors or by a qualifying floating charge holder (as defined in paragraph 14 of Schedule B1 to IA 1986).
- 2.3 No receiver or manager (which expression shall include an administrative receiver) has been appointed in respect of any or all of the Target's assets, nor has any power of sale or power to appoint a receiver or manager under the terms of any mortgage, charge or other security in respect of all or any assets of the Target become exercisable.
- 2.4 The Target is not insolvent or unable to pay its debts (within the meaning of section 123 IA 1986), nor has it stopped paying its debts as they fall due (otherwise than by reason of a bona fide dispute as to their amount or enforceability).
- 2.5 No distress, execution or other process has been levied on any asset of the Target by any of its creditors.
- 2.6 No voluntary arrangement under section 1 IA 1986 or scheme of arrangement under section 895 CA 2006 in respect of the Target or any other compromise or arrangement in respect of the Target's creditors generally, or any class of them, has been proposed or adopted.
- 2.7 No statutory demand has been served on the Target, which has not been paid in full or been withdrawn.
- 2.8 The Target has not suspended or ceased or threatened to suspend or cease to carry on all or a material part of its business.
- 2.9 The Target has not been a party to any transaction at an undervalue (as defined in section 238 IA 1986), nor has it given or received any preference (as defined in section 239 IA 1986) in either case

within the period of two years ending on the date of this Agreement, nor has the Target at any time been a party to any transaction defrauding creditors (as defined in section 423 IA 1986).

#### 3. ACCOUNTS

#### 3.1 General

- 3.1.1 The Accounts give a true and fair view of the state of affairs of the Target as at the Accounts Date.
- 3.1.2 The Accounts:
  - (a) were prepared and audited in accordance with:
    - (i) the laws of England and Wales, including the CA 2006; and
    - (ii) generally accepted accounting standards in England and Wales as were in force on the Accounts Date; and
  - (b) were prepared using the same accounting standards used in the preparation of the Target's audited accounts for the preceding two (2) years.

#### 3.2 Management Accounts

The Management Accounts: (i) were prepared in accordance with US GAAP; and (ii) show with reasonable accuracy (taking account of the purpose for which they have been prepared), the Target's trading income and expenditure for the period in respect of which they were prepared, on a basis consistent with the preparation of previous management accounts of the Target.

## 3.3 Business since the Accounts Date

Since the Accounts Date:

- (a) the Target has carried on its business in the ordinary and usual course;
- (b) no distribution of capital or income has been declared, made or paid by the Target;
- (c) the Target has not acquired or disposed of, nor agreed to acquire or dispose of, any asset otherwise than in the ordinary and usual course of trading;
- (d) the Target has not assumed or incurred, nor agreed to assume or incur, any liability or obligation otherwise than in the ordinary and usual course of trading or other than as taken into account in the Management Accounts;
- (e) no resolution of the Target's shareholders has been passed (except for those representing the ordinary business of an annual general meeting); and
- (f) the Target has not repaid or redeemed any share or loan capital.

## 3.4 Completion Statement

The Estimated Completion Statement has been prepared in good faith and in accordance with the accounting principles and policies set out in Part C to Schedule 6.

#### 4. TARGET ASSETS

### 4.1 Ownership

- 4.1.1 Save as disclosed in the Disclosure Letter, all material assets (other than the LPR Assets) reasonably necessary for the operation of the Business, as currently carried on, are legally and beneficially owned by the Target, free from Encumbrances other than liens arising in the ordinary course of business. Such assets, together with the LPR Assets, constitute all assets that are reasonably necessary for the operation of the Business, as currently carried on.
- 4.1.2 The Target's fixed asset register as at 31 May 2015 is annexed to the Disclosure Letter and sets out a materially accurate record of all material plant, machinery, vehicles and equipment owned or used by it.

## 4.2 Possession

All of the material assets owned by the Target or in respect of which the Target have a right of use which are capable of possession are in the possession or under the control of the Target.

## 4.3 Condition and Adequacy

All material plant, machinery and equipment used by the Target is in satisfactory condition and is capable of being used for the purpose for which it is currently being used by the Target.

# 4.4 Hire purchase and leased assets

Copies of any material hiring or leasing agreement, hire purchase agreement, or any other similar agreement to which the Target is currently a party are annexed to the Disclosure Letter.

#### 4.5 Debts

- 4.5.1 The Target is not currently party to a legally binding agreement under which it has factored, sold or discounted any of its debts.
- 4.5.2 The Target is not currently legally bound by an agreement under which it has granted credit terms exceeding 120 days to its creditors.

### 4.6 Insurance

- 4.6.1 A list of all current insurance policies in respect of which the Target has a current interest is annexed to the Disclosure Letter (the "**Policies**"). Such list is complete and accurate in all material respects.
- 4.6.2 All premiums in respect of the Policies have been duly paid to date and, so far as the Parent is aware, the Policies are in force and not voidable.
- 4.6.3 A list of all material claims outstanding under the Policies as at the Completion Date is annexed to the Disclosure Letter and, so far as the Parent is aware, such list is complete and accurate.

#### 5. LPR ASSETS

5.1 Immediately prior to Completion, the LPR Assets were legally and beneficially owned by the Parent free from all Encumbrances. The Parent is not a party to any transaction pursuant to or as a result of which any of the LPR Assets is liable to be transferred to another person. As far as the Parent is

aware, upon Completion the Purchaser will become the legal and beneficial owner of the LPR Assets, free from all Encumbrances.

5.2 The LPR Assets are the only assets that are currently used in the Business but not owned by the Target.

#### 6. LIABILITIES

### 6.1 Facilities

Details of all overdrafts, loans or other financial facilities currently outstanding or available to the Target (other than trade indebtedness) are contained in the Disclosure Letter and, so far as the Parent is aware, nothing has been done or omitted to be done whereby the continuance of any such facilities in full force and effect will be prejudiced (save where the prepayment of such facilities is expressly provided for in the Transaction Documents).

#### 6.2 Liabilities since Accounts Date

Except as set forth in the Accounts, there are no liabilities of the Target other than liabilities incurred in the ordinary course of business after the Accounts Date.

#### 6.3 Guarantees and indemnities

There is not outstanding any guarantee, indemnity, security, bond, letter of comfort or other similar obligation given by or for the benefit of the Target.

#### 6.4 Events of default

No written notice (which is still current) has been received by the Target to the effect that the Target is in material default under the terms of any borrowing made by it.

## 6.5 Reorganisation

Except for the Transaction Documents and the documents to be delivered to the Purchaser pursuant to paragraph 1.1(b)(viii) of Schedule 3, there are no other agreements or documents relating to the Reorganisation or otherwise providing for the transfer of assets from or liabilities to the Target and entered into with the Seller or any of its Affiliates (other than the Target) in the six (6) months preceding the Completion Date.

#### 7. TRADING ARRANGEMENTS

#### 7.1 Contracts

- 7.1.1 During the last six months, no Material Supplier has ceased supplying the Target nor during the last six months has any Material Supplier notified the Target in writing that it will either cease supplying the Target or materially reduce its supplies to the Target or materially increase its prices.
- 7.2 During the last six months, no Material Customer has ceased trading with the Target nor during the last six months has any Material Customer notified the Target in writing that it will either cease trading with the Target or materially reduce its trade with the Target.



- 7.3 Materially complete and accurate copies of all Material Contracts have been disclosed to the Purchaser. The Material Contracts are the only agreements that are material to the business of the Target and the Business as it is currently conducted.
- 7.4 The Target is not in breach of any Material Contract as at the date of this Agreement.

#### 8. COMPLIANCE AND LITIGATION

#### 8.1 Compliance with statutes

- 8.1.1 During the last three years the Target has complied, and currently complies, in all material respects with all applicable statutes and regulations of the United Kingdom and any foreign jurisdiction in which the Business is carried on, non-compliance with which would have a more than immaterial effect upon the Target or the Business.
- 8.1.2 The Target has in its possession or under its direct or indirect control all material documentation required by law to be disclosed to a regulator with authority over the Target in connection with the manufacture, storage or testing of goods manufactured and supplied by the Target.

# 8.2 Licences and consents

- 8.2.1 Other than in respect of Intellectual Property Rights and/or the Properties, the Disclosure Letter includes details of all material licences, consents, approvals, permissions, permits, certificates, qualifications, registrations and other authorisations (public and private) legally required for the operation of the Business in the places and in the manner in which the Business is now carried on (together the "Material Consents").
- 8.2.2 The Target has all Material Consents and all such Material Consents are in full force and effect and have been complied with in all material respects.

### 8.3 Litigation

8.3.1 Apart from debt collection in the normal course of trading, the Target is not involved in any ongoing material civil, criminal, arbitration or administrative proceedings in the UK or any other jurisdiction ("**Proceedings**").

8.3.2 None of the Target, the Vendor or the Parent has received written notice that any Proceedings are currently pending against the Target.

### 9. INTELLECTUAL PROPERTY AND INFORMATION TECHNOLOGY

### 9.1 Material intellectual property rights

- 9.1.1 The LPR IP constitutes all material Intellectual Property Rights (and applications for such rights) which were owned by the Target, the Parent or any Vendor Group Company immediately prior to Completion and which primarily relate to the Business.
- 9.1.2 None of the Target, the Parent or any Vendor Group Company has received written notice to indicate that any of the LPR IP is being challenged or attacked by any third party or by any relevant registry and all fees due and payable in respect of the LPR IP have been paid.

#### 9.2 IPR Agreements

- 9.2.1 Details of all IPR Agreements which are material to the Business have been disclosed to the Purchaser.
- 9.2.2 So far as the Parent is aware, no party to any IPR Agreement which is material to the Business is in material default and none of the Target, the Parent or any member of the Vendor Group has been notified in writing of any material disputes relating to or arising out of any IPR Agreement which is material to the Business.

## 9.3 Infringements

- 9.3.1 So far as the Parent is aware, none of the LPR IP is being infringed.
- 9.3.2 None of the Target, or any member of the Vendor Group have received written notice to indicate that the Target or (to the extent that the alleged infringement relates to the Business) any member of the Vendor Group is currently infringing any Intellectual Property Rights of any other person.

## 9.4 Confidential Information

So far as the Parent is aware, all Confidential Information material to the Business has been kept confidential by the Parent, the Target and all other members of the Vendor Group and has not been disclosed otherwise than in the ordinary course of business subject to an obligation of confidentiality on the person to whom it was disclosed.

#### 9.5 Information Technology

- 9.5.1 The IT Systems are in satisfactory working order in all material respects and there has not been a failure of the IT Systems during the two years prior to the date hereof which had a materially disruptive effect on the Target's ability to carry on its business in the ordinary course of trading for a significant period of time.
- 9.5.2 The IT Systems are the only IT systems that are required in order to operate the Business as currently operated.

#### 10. OFFICERS AND EMPLOYEES

#### 10.1 Particulars

- 10.1.1 The persons named as such in Schedule 1 are the only persons currently appointed to the board of directors of the Target.
- 10.1.2 The particulars shown in the schedule of employees annexed to the Disclosure Letter discloses, in relation to the employees (or, where appropriate, to each category of employee) of the Target (the "Employees") their job function, salary and notice period.
- 10.1.3 The Target is not under a legal obligation to make any future change in the remuneration or benefits of any of the Employees other than salary or wage increases in the ordinary course of business.
- 10.1.4 No Employee will become entitled to receive a payment by virtue of Completion of this Agreement.

#### 10.2 Compliance

The Target has complied in all material respects with all its statutory obligations to or in respect of the Employees arising out of their terms and conditions of employment and no amount due to or in respect of any Employee is in arrear and unpaid other than salary and expenses for the month current at the date of this Agreement.

#### 10.3 Notice

No Employee has given written notice to his employing Target to terminate his contract of employment or is under written notice of dismissal.

# 11. PENSIONS

11.1 In this paragraph 11:

"Relevant Benefits" means any benefits that are provided under a pension scheme (as defined in section 150(1) Finance Act 2004).

"Relevant Person" means any past or present employee, officer or director of the Target.

- 11.2 Save as specifically set out in the Disclosure Letter, the Target is not a party to, or participates in or contributes to any scheme, agreement or arrangement for the provision of any Relevant Benefits for or in respect of any Relevant Person.
- 11.3 All contributions and other amounts which have fallen due for payment by the Target have been paid.
- 11.4 Details of the benefits payable in respect of the employees on their death in service or incapacity have been provided to the Purchaser; all lump sum benefits payable in the event of death and incapacity benefits are insured, and all premiums payable under such insurance contracts have been paid.
- 11.5 The Target has complied with the applicable requirements of the Welfare Reform and Pensions Act 1999 relating to the provision of stakeholder pensions.
- 12. PROPERTY
- 12.1 Interpretation

In this Paragraph 12:

- (a) each Warranty which is expressed to be given in relation to the "**Property**" shall be deemed to be given in relation to each of the Properties as if it had been repeated with respect to each of the Properties and each and every part of it; and
- (b) reference to the "**Owner**" shall, where the context so admits, be a reference to the Target which is the owner of the relevant Property as shown in Schedule 8.

### 12.2 Warranties

Schedule 8 contains a complete and accurate list of the properties owned, controlled, used or occupied by the Target and the information given in Schedule 8 is accurate.

#### 13. ENVIRONMENT HEALTH & SAFETY

13.1 In this Paragraph 13:

"Dangerous Substance" means any natural or artificial substance (alone or in combination) likely to cause significant damage to the Environment;

"Environment" means any or all of the following media (alone or in combination): air, water, soil and/or ecosystem;

"EHS Law" means all statutes and regulations concerning the protection of the Environment or the spillage, treatment or disposal of a Dangerous Substance or health and safety at work (excluding any statutes or regulations which relate to town and country planning) which are in force as at the date of this Agreement and which are applicable to and legally binding upon the Target;

"EHS Licences" means any permit, licence, authorisation, consent or other approval required under EHS Law;

"Regulatory Authority" means any governmental authority, agency or department having authority under, or jurisdiction in respect of, any EHS Law.

- 13.2 The Target has all EHS Licences which are material to and necessary for its business as at the date of this Agreement.
- 13.3 The Target is in material compliance with EHS Law.
- 13.4 There are currently no formal legal proceedings relating to a breach of EHS Law currently being brought (or, as far as the Parent is aware, threatened to be brought) by a Regulatory Authority against the Target.

#### 14. OBLIGATIONS TOWARDS THE PARENT AND THE VENDOR

As of Completion, there are no liabilities or contracts between the Target on the one hand, and the Parent or the Vendor or their respective Affiliates (other than the Target) on the other hand.

### Part C - Tax Warranties

### 1. TAX

## 1.1 Fundamental Tax Warranties

- 1.1.1 Provision or reserve (as appropriate) has been made in the Accounts in accordance with generally accepted accounting practice for all Tax, including deferred tax, liable to be assessed on the Target or for which the Target is accountable in respect of all income, profits or gains earned, accrued or received on or before the Accounts Date or deemed to have been or treated for Taxation purposes as earned, accrued or received on or before the Accounts Date, and in respect of any act, omission, transaction or other event occurring or deemed to have occurred on or before the Accounts Date.
- 1.1.2 No Tax has, may have arisen or will arise to the Target (or would arise or have arisen but for the use of any available reliefs) in respect of or in connection with any income, profits or gains earned, accrued or received since the Accounts Date and on or before Completion or deemed to have been or treated for Taxation purposes as earned, accrued or received since the Accounts Date and on or before Completion or in respect of or in connection with any act, omission, transaction or other event occurring or deemed to have occurred since the Accounts Date and on or before Completion, other than in respect of normal trading income arising in the ordinary course of the Target's business.
- 1.1.3 The Target is not, and will not become in respect of periods, part periods or events falling or occurring on or before Completion, liable to pay, or to pay any amount in respect of or make reimbursement or indemnity in respect of, any Tax (or amounts corresponding to any Tax) which is primarily chargeable on or attributable to any other person. The Target is not liable as an agent, permanent establishment or prescribed person for any Tax liability of another person.
- 1.1.4 The Target has properly paid all Tax prior to Completion which it has become liable to pay prior to Completion.
- 1.1.5 No charge to Tax will arise on the Target as a result of entry into or completion of this Agreement.

#### 1.2 General

- 1.2.1 The Target is not and has not been party to, any transaction or arrangement under which it may be or has been required to compute its profits or losses for tax purposes as if arm's length terms had been made or imposed instead of the actual terms, or otherwise to make any adjustment for tax purposes to the terms on which the transaction or arrangement took place. The Target has sufficient information and records to enable it to comply with, or establish that it is not subject to the operation of Part 4 TIOPA 2010 or any similar legislation in any jurisdiction outside the United Kingdom.
- 1.2.2 The Target is not and has never been a close company within the meaning of section 439 CTA 2010.
- 1.2.3 All payments by the Target to any person which ought to have been made under deduction or withholding of Tax have been so made and the Tax so deducted or withheld has been properly and punctually accounted to the relevant Taxation Authority.

## 1.3 Returns and compliance

1.3.1 The Target has made all returns required for Tax purposes and none of such returns is being or, so far as the Parent is aware, will be disputed by HMRC or other Taxation Authority.



- 1.3.2 The Target has prepared, kept and preserved full, sufficient, accurate and up-to-date records to enable it to make and complete returns for Tax purposes and to calculate the liability to Tax or the amount of a relief arising on the disposal of any asset owned at the Accounts Date or acquired since the Accounts Date but before Completion and otherwise as required by Law.
- 1.3.3 Neither the Target, nor any of its directors or officers in their capacity as such, have within the last six (6) years paid or become liable to pay, nor are there any circumstances which may cause any of them to become liable to pay, any penalty, fine, surcharge or interest in connection with Tax.
- 1.3.4 No Taxation Authority has agreed to operate any special arrangement (being an arrangement which is not based on a strict and detailed application of the relevant legislation, statements of practice or published extra-statutory concessions) in relation to the Target's affairs.
- 1.3.5 The Target is not involved in any dispute with any Taxation Authority and has not within the last six (6) years been subject to any visit, audit, investigation, discovery or access order by any Taxation Authority. The Parent is not aware of any circumstances existing which make it likely that a visit, audit, investigation, discovery or access order will be made.

### 1.4 VAT

- 1.4.1 The Target has, throughout the whole of the period beginning three years before the Accounts Date and ending on the date hereof, been registered and been eligible to be registered and is a taxable person for the purposes of the VATA 1994 and such registration is not subject to any conditions imposed by or agreed with any Taxation Authority. The Target has never been a member of a group of companies for the purposes of sections 43 to 43C VATA 1994.
- 1.4.2 The Target has complied with all Laws relating to VAT, has made and accounted for all payments of VAT due from it, has made and delivered all required VAT returns to applicable Taxation Authorities, is not liable to pay any penalties for non-payment of VAT and has not been required to give security in respect of VAT.
- 1.4.3 The Target does not owns any assets which are capital items subject to the capital goods scheme under Part XV of the Value Added Tax Regulations 1995, nor has exercised any option to tax under Part 1 of Schedule 10 to VATA 1994.

# 1.5 Groups of companies

- 1.5.1 Save as expressly set out in the Disclosure Letter, there are no claims to or surrenders of group relief (within the meaning of Part 5 CTA 2010) by or to the Target which have yet to be finally agreed or determined, and no payments for group relief (within the meaning of Part 5 CTA 2010) by or to the Target which remain outstanding or could be reduced or increased.
- 1.5.2 The Target is not, and has not within the last six years been, party to any such arrangement as is mentioned in section 36 Finance Act 1998 (group payment arrangements).

### 1.6 Employees

- 1.6.1 The Target has complied fully with its legal obligations relating to PAYE and National Insurance contributions and any similar amounts payable to a Taxation Authority outside the United Kingdom.
- 1.6.2 No Tax has arisen or is likely to arise to the Target as a result of any person acquiring, holding or disposing of shares or securities or an interest in shares or securities where the right or opportunity to acquire the same is or was available by reason of employment.

iss152795_ex2-1.htm 152795 - FINAL	08/11/2015 03:01 PM	56 of 78
------------------------------------	---------------------	----------

- 1.6.3 No relevant step (within the meaning of Part 7A ITEPA 2003) has been taken in pursuance of, or in connection with, arrangements concerned with the position of rewards or recognition or loans in connection with any employee or former employee (or any associate of such persons) of the Target.
- 1.6.4 Any amounts paid by the Target to, or for the direct or indirect benefit of, a person who is or who may be regarded by any Taxation Authority as an employee of the Target, or who would be regarded as such an employee but for the involvement of an intermediary company, has been made to that person directly and not to any company or other entity associated with that person.
- 1.6.5 No Tax has arisen or is likely to arise to the Target by reason of any legal provision relating to managed service companies (as defined in section 61B ITEPA 2003).

## 1.7 Stamp taxes

There is no document or instrument to which the Target is a party, or which is necessary to establish the Target's rights in, or title to, any asset, which is or could become liable to stamp duty (or any similar duty or tax in a jurisdiction outside the United Kingdom) which has not been duly stamped or in respect of which the relevant duty or Tax has not been paid.

# 1.8 International

The Target was incorporated in, and is and always has been resident only in, its country of incorporation for Tax purposes and for the purposes of any double taxation agreement. The Target is not liable for, and has at no time incurred any, Tax in any jurisdiction other than its jurisdiction of incorporation.

#### **SCHEDULE 5**

#### Limitations

## 1. MAXIMUM LIABILITY

- 1.1 Subject to the provisions of paragraph 1.2 below, the total aggregate liability of the Parent and the Vendor in respect of any breach of, or claim under, this Agreement (other than claims for a price adjustment under clause 7, to which the limitation set out in clause 7.3.2 shall apply) shall not exceed a sum equal to \_\_% of the Price; and
- 1.2 Notwithstanding the provisions of paragraph 1.1 above, the total aggregate liability of the Parent in respect of any breach of, or claim under, any of the Fundamental Warranties (excluding the Fundamental Tax Warranties) shall not exceed a sum equal to the Price.

### 2. SOLE RECOURSE TO ESCROW

The total aggregate liability of the Parent and, where applicable, the Vendor in respect of any claim arising under this Agreement shall be limited in accordance with clause 6.1.2 (other than claims for a price adjustment under clause 7, to which the provisions of clause 7.3 shall apply).

#### **3.** WARRANTY CLAIM THRESHOLDS

- 3.1 The Parent shall have no liability in respect of a Warranty Claim unless:
  - (a) the liability under this Agreement in respect of that Warranty Claim (when aggregated with the liabilities in respect of all other Warranty Claims concerning the same or substantially similar subject matter) exceeds \$\_\_\_\_\_, in which case the Purchaser shall be entitled to claim the whole of such sum and not merely the excess; and
- 3.2 For the purpose of paragraph 3.1 above, the liability in respect of a Warranty Claim under this Agreement shall exclude any liability for costs, expenses and interest incurred in investigating and pursuing such Warranty Claim.

#### 4. TIME LIMITS

- 4.1 The Parent shall have no liability in respect of a Warranty Claim (other than a claim for a breach of any Fundamental Warranty) unless the Purchaser shall have given notice in writing to the Parent of such Warranty Claim specifying (in reasonable detail in so far as is known to the Purchaser) the matter which gives rise to the claim, the nature of the claim and the estimated amount claimed in respect thereof (including details of the Purchaser's calculation of such estimated amount), such notice to be given not later than the date 12 months after the Completion Date.
- 4.2 Save as may be expressly agreed in writing to the contrary on a case-by-case basis by the Purchaser and the Parent, all liability in respect of a Warranty Claim notified to the Parent in accordance with paragraph 4.1 above shall (if such Warranty Claim has not previously been satisfied, settled or

withdrawn) be extinguished on the expiry of six months from the date of such notification of the Warranty Claim unless the Purchaser shall within such period have issued and validly served legal proceedings in respect of such Warranty Claim on the Parent.

#### 5. MATTERS REFLECTED IN THE ACCOUNTS

#### 5.1 Allowances, provisions and reserves

The Parent shall not be liable in respect of a Warranty Claim to the extent that allowance, provision or reserve in respect of the matter giving rise to the Warranty Claim or a specific category of such matters is explicitly made in the Accounts or in the Completion Statement or such matter is otherwise taken explicitly into account or reflected in the preparation of the Accounts or of the Completion Statement.

#### 5.2 Excess provisions, overvalues, undervalues and written-off amounts

- 5.2.1 If and to the extent that:
  - (a) the amount of any allowance, provision or reserve other than any provision or reserve for deferred Taxation made in the Accounts or in the Completion Statement or otherwise taken into account or reflected therein is found to be in excess of the matter for which such allowance, provision or reserve was made;
  - (b) any asset is found to have been included at an undervalue in the Accounts or in the Completion Statement or any liability is found to have been included at an overvalue in Accounts or in the Completion Statement; or
  - (c) any sum is received by the Target which has previously been written off as irrecoverable in the accounts (and accordingly not included as an asset in the Accounts or in the Completion Statement), then the amount of any such excess, undervalue, overvalue or receipt as the case may be (the "**Excess Amount**") shall be credited against and applied in relieving any liability for which the Parent may otherwise be liable in respect of any Warranty Claims.
- 5.2.2 The Parent shall not be liable for any Warranty Claim to the extent that any allowance, provision or reserve made in the Accounts or in the Completion Statement in respect of the fact, matter, event or circumstance giving rise to such claim is insufficient by reason of any change to legislation, any increase in rates of Taxation or any change in the published practice of a Tax Authority, in each case made on and/or after Completion with retrospective effect.

#### 6. CHANGES OF LAW

The Parent shall have no liability in respect of any Warranty Claim to the extent that it arises or is increased as a result of any change in Law (or change in the interpretation of Law) or in

administrative practice of any Authority occurring after the date of this Agreement (whether or not the change purports to be effective retrospectively in whole or in part).

#### 7. ACCOUNTING BASES ETC.

The Parent shall not be liable in respect of any Warranty Claim to the extent that it arises or is increased as a result of any change after Completion in the accounting bases, policies, practices or methods applied in preparing any accounts or valuing any assets or liabilities of the Target from those used in preparing the Accounts.

#### 8. ACTS OF PURCHASER

- 8.1 The Parent shall not be liable in respect of a Warranty Claim to the extent such liability is attributable to:
  - (a) any act, omission or transaction carried out on or after Completion by or at the request of or with the consent of the Purchaser or any of its Affiliates;
  - (b) any cessation of, or any material change in, the nature or conduct of any business carried on by the Purchaser or the Target, occurring on and/or after Completion (other than as a means of mitigating a Loss);
  - (c) the making of a claim, election, surrender or disclaimer or the giving of a notice or consent after Completion by the Purchaser or any member of the Enlarged Group or any other person connected with any of them (unless any provision or reserve for Taxation in the Accounts is computed on the basis that such action will be taken);
  - (d) anything expressly provided to be done or omitted to be done pursuant to this Agreement or any other Transaction Document; or
  - (e) anything done before Completion at the request of the Purchaser or any of its then Affiliates.
- 8.2 The Parent shall not be liable in respect of a Warranty Claim under Tax Warranties to the extent such liability is attributable to or is increased as a result of a voluntary act, transaction or omission carried out by the Purchaser, the Target or entity connected with the Purchaser after Completion being an act which:
  - (a) is not in the ordinary course of business of the Target as such business was conducted at Completion; and
  - (b) the Target was not legally committed to do under a commitment that existed on or before Completion.

### 9. COVERED LOSSES

9.1 The Parent shall not be liable in respect of any Warranty Claim to the extent that the claim relates to a Covered Loss.

## 9.2 A "Covered Loss" means:

(a) any Loss which is covered by a policy of insurance in force for the benefit of the Target at Completion and payment is made under such policy, less all reasonable costs, charges and



expenses incurred in effecting recovery including for the avoidance of doubt any additional costs, charges or expenses (including any increase in premium) relevant to the ongoing provision of any such policy of insurance as are occasioned by reason of the making of such claim or claims to recovery; and

(b) any Loss in respect of which the Purchaser or any member of the Enlarged Group has received compensation from any person (whether under any provision of law, contract or otherwise howsoever) other than from any Vendor Group Company under this Agreement or other Transaction Documents.

## **10.** SUBSEQUENT RECOVERY

If the Purchaser recovers an amount from the Parent in respect of any Warranty Claim and the Purchaser or any member of the Enlarged Group subsequently recovers or receives from any other person any amount or benefit in respect of the matter giving rise to the Warranty Claim, the Purchaser shall thereupon re-pay to the Parent at the direction of the Parent the value of the amount or benefit so recovered or received (except to any extent that the liability of the Parent in respect of the Warranty Claim was previously reduced to take account of such amount or benefit). Any amount so re-paid to the Parent shall be apportioned between them pro-rata as to their respective contributions to the original payment to the Purchaser in respect of the Warranty Claim.

## 11. **Related benefits**

In assessing any damages or other amounts payable in respect of a Warranty Claim there shall be taken into account any related savings or other net benefits which are actually enjoyed by any member of the Enlarged Group. In assessing any damages or other amounts payable in respect of a Warranty Claim that relates to Tax there shall be taken into account any Relief arising from the Tax that is the subject of such Warranty Claim.

### 12. CONTINGENT AND NON-QUANTIFIABLE LIABILITIES

The Parent shall not be liable in respect of any Warranty Claim to the extent that it relates to a liability which is contingent or not capable of being quantified unless and until the liability ceases to be contingent or becomes capable of being quantified, as the case may be, save that, where notice of such a Warranty Claim has been given to the Parent in accordance with paragraph 4.1 of this Schedule 5, and therein expressly identified as being a contingent and/or claim which is then incapable of being quantified, then the applicable time period for the commencement of legal proceedings in respect of such Warranty Claim under paragraph 4.2 of this Schedule 5 shall be postponed so as to thereafter commence on the date on which such Warranty Claim first ceases to be so contingent and is reasonably capable of being quantified (provided always that legal proceedings in respect of such Warranty Claim are in any event commenced no later than 18 months after the date of Completion).

#### **13.** NO RECOVERY FOR CONSEQUENTIAL LOSS

The Parent shall not be liable in respect of any Warranty Claim for any Consequential Loss.

#### 14. PURCHASER'S KNOWLEDGE

The Parent shall not be liable in respect of any Warranty Claim to the extent that the Purchaser or any of its Affiliates (or any of the officers or employees of any of the foregoing) actually knew (with

sufficient detail so as to be reasonably able to identify the nature and scope of the relevant matters) of the matters giving rise to the claim at the date of this Agreement.

## 15. MITIGATION

The Purchaser shall take (and shall procure that all other relevant members of the Enlarged Group shall take) all reasonable steps to mitigate any loss or liability which is or might become the subject of a Warranty Claim (including utilizing any Relief that is available to the Company to reduce any Tax that is the subject of such Warranty Claim).

## 16. NO DUPLICATION OF RECOVERY

- 16.1 The Parent shall not be liable in respect of any Warranty Claim to the extent the Purchaser or member of the Enlarged Group has already recovered or received from any other person any amount or benefit in respect of the matter giving rise to the Warranty Claim.
- 16.2 The Purchaser shall not be entitled to recover damages or otherwise obtain reimbursement or restitution more than once in respect of the same Loss.

# **SCHEDULE 6**

# **Completion Statement**

# Part A - Pro-Forma Completion Statement

iss152795	ex2-1.htm

152795 - FINAL

# Part B - Estimated Completion Statement

iss152795	ex2-1.htm

152795 - FINAL

# Part C - Completion Statement accounting policies

# **SCHEDULE 7**

# 2013 Audited Accounts

[attached behind this page]

# IMAGE SENSING SYSTEMS UK LIMITED

# ABBREVIATED ACCOUNTS

# FOR THE YEAR ENDED 31 DECEMBER 2013

Company Registration Number

# **SCHEDULE 8**

# **Real Properties**

# Part A - Owned Properties

Part B - Leased Properties

# **SCHEDULE 9**

LPR Assets

# Part A - LPR Accounts Receivable

# Part B - LPR Inventory

iss152795	ex2-1.htm

# Part C - LPR IP

# PATENTS

# **REGISTERED DESIGNS**

# TRADE MARKS

# **Registered Trade Marks**

# **Unregistered Trade Marks**

iss152795	ex2-1.htm

# DOMAIN NAMES

iss152795	ex2-1.htm

# Part D - Customer List and Non-Compete

SIGNATURE PAGE TO SALE A	AND PURCHASE AGREEMENT
IN WITNESS THIS DEED has been executed by the Parties and is intended to	be and is delivered on the date appearing on its first page.
EXECUTED as a DEED by TAGMASTER AKTIEBOLAG	
acting by its duly authorised officer	/s/
	signature of authorised officer
in the presence of:	Print name of authorised officer /s/
	Signature of witness
	Print name of witness
	Address of witness:

08/11/2015 03:01 PM

76 of 78

152795 - FINAL

iss152795\_ex2-1.htm

# **EXECUTED** as a **DEED** by

# IMAGE SENSING SYSTEMS, INC.

acting by its duly authorised officer

in the presence of:

/s/\_\_\_\_\_ signature of authorised officer Print name of authorised officer /s/\_\_\_\_\_ Signature of witness

Print name of witness

Address of witness:

# **EXECUTED** as a **DEED** by

# IMAGE SENSING SYSTEMS EMEA LIMITED

acting by its duly authorised officer

in the presence of:

signature of authorised office Print name of authorised office /s/	Print name of authorised	
/s/ Signature of witness Print name of witness	/s/	offic
Signature of witness Print name of witness		
Print name of witness	Signature of witness	
	5	
Address of witness:	Print name of witness	
	Address of witness:	

# Exhibit 31.1

# CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Dale E. Parker, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Image Sensing Systems, Inc. for its fiscal quarter ended June 30, 2015;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 11, 2015

/s/ Dale E. Parker Name: Dale E. Parker Title: Interim President and Interim Chief Executive Officer

# Exhibit 31.2

# CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Dale E. Parker, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Image Sensing Systems, Inc. for its fiscal quarter ended June 30, 2015;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 11, 2015

/s/ Dale E. Parker Name: Dale E. Parker Title: Chief Financial Officer EX-32.1 152795 - FINAL

# Exhibit 32.1

# CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Image Sensing Systems, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2015, as filed with the Securities and Exchange Commission (the "Report"), I, Dale E. Parker, Interim President, Interim Chief Executive Officer and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Dale E. Parker Dale E. Parker Interim President, Interim Chief Executive Officer and Chief Financial Officer August 11, 2015